52-1306

ORNL-TM-4663

CONDOS—A Model and Computer Code to Estimate Population and Individual Radiation Doses to Man from the Distribution, Use, and Disposal of Consumer Products that Contain Radioactive Materials

> F. R. O'Donnell L. R. McKay O. W. Burke F. H. Clark

MASTER

Environmental Sciences Division Publication No. 638



BLANK PAGE

· · •

•

Printed in the United States of America. Available from National Technical Information Service U.S. Department of Commerce 5285 Port Royal Road, Springfield, Virginia 22161 Price: Printed Copy \$5.45; Microfiche \$2.25

This report was prepared as an account of work sponsored by the United States Government. Neither the United States nor the United States Atomic Energy Commission, nor any of their employees, nor any of their contractors, subcontractors, or their employees, makes any warranty, express or implied, or assumes any legal liability or responsibility for the accuracy, completeness or usefulness of any information, apparatus, product or process disclosed, or represents that its use would not infringe privately owned rights.



ORNL-TM-4663 UC-41 — Health and Safety

Contract No. W-7405-eng-26

CONDOS--A MODEL AND COMPUTER CODE TO ESTIMATE POPULATION AND INDIVIDUAL RADIATION DOSES TO MAN FROM THE DISTRIBUTION, USE, AND DISPOSAL OF CONSUMER PRODUCTS THAT CONTAIN RADIOACTIVE MATERIALS

F. R. O'Donnell and L. R. McKay Environmental Sciences Division

O. W. Burke and F. H. Clark Instrumentation and Controls Division

Environmental Sciences Division Publication No. 638

MAY 1975

NOTICE This document contains information of a preliminary nature and was prepared primarily for internal use at the Oak Ridge National Laboratory. It is subject to revision or correction and therefore does not represent a final report.

MASTER

OAK RIDGE NATIONAL LABORATORY Oak Ridge, Tennessee 37830 operated by UNION CARBIDE CORPORATION for the U. S. ENERGY RESEARCH AND DEVELOPMENT ADMINISTRATION

DISTRIBUTION OF THIS DOCUMENT UNLIMITED ()

TABLE OF CONTENTS

				ļ	Page
ABS	TRACT		•	•	١
1.	INTR	DDUCTION	•	•	2
2.	THE I	MODEL		•	4
	2.1	Product Description	•	•	4
	2.2	Model of Product Life Span and Conditions of Exposure	•	•	5
	2.3	Methodology for Estimation of Radiation Doses to Man	•	•	11
	2.4	Discussion	•	•	15
3.	MATH	EMATICAL MODELS OF THE DOSE CALCULATIONS	•	•	17
	3.1	External Exposure	•		18
		3.1.1 Dose from direct irradiation	•	•	18
		3.1.2 Dose from immersion in contaminated air	•	•	23
		3.1.3 Dose to skin from contact with a beta-particle			
		source	•	•	24
	3.2	Internal Exposure	•	•	26
		3.2.1 Dose from ingestion	•	•	26
		3.2.2 Dose from inhalation	•	•	27
4.	THE	COMPUTER CODE	•	•	28
	4.1	The Computer Solution	•	•	28
	4.2	Auxiliary Calculations	•	•	30
		4.2.1 Finite volume correction factor	•	•	30
		4.2.2 Concentration of radioactive material it air .	•	•	31
		4.2.3 Geometric conversions	•	•	32
	4.3	Logical Flow of the Code	•		34
	4.4	Description of Subroutines	•		34

BLANK PAGE

TABLE OF CONTENTS (cont'd)

					Page
	4.5	Definition of Program Variables	• • •	• • • •	40
	4.6	Description of Input	• •		52
	4.7	Description of Output			55
5.	SUMM	IMARY AND FUTURE DEVELOPMENTS			55
6.	ACKN	NOWLEDGMENTS			58
REF	ERENC	ICES		• • • •	59
APP	ENDIX	X A			
D	ATA B	BASE (JULY 1974)	• • •		61
APP	ENDIX	ХВ			
D	ERIVA	ATION OF THE MODEL EQUATION FOR CALCULATING THE	ε βετά	L	
	DOSE	SE RATE	• • •		71
APP	ENDIX	IX C			
S	AMPLE	E PROBLEM	• • •	• • • •	77
APP	ENDIX	IX D			
L	ISTIN	ING OF THE CONDOS COMPUTER CODE	• • •	• • • •	87

LIST OF TABLES

Number		Page
1	List of components to be considered in modeling	
	the life span of a consumer product	7
A.1	Data variables related to individual	
	radionuclides	63
A.2	Photon yields per disintegration of each	
	radionuclide (J) per energy interval (I)	64
A.3	Beta particle yields per disintegration of	
	each radionuclide (J) per energy interval (I)	67
A.4	Data variables related to photon and beta	
	particle energy	70
C.1	Life span model for hypothetical tableware	78
C.2	List of input variables that describe the events	
	in the life span of the hypothetical tableware .	79
C.3	Computer output for sample problem hypothetical	
	tableware	82

LIST OF FIGURES

Number		Page
1	Outline of model of the life span of a consumer product	6
2	Flow diagram of the methodology for estimation of	
	radiation doses to man from a consumer product	
	containing radioactive material	13
3	Coordinate system for conversion from point to	
	disk source	33
4	Logical flow diagram of the main program	35
5	Logical flow diagram of the subroutine INPUT	37
6	Logical flow diagram of subroutine PREFAS	38
7	Logical flow diagram of subroutine OUTPUT	39
8	Logical flow diagram of subroutine CALC	41
9	Logical flow diagram of subroutine F(ABS)	42
10	Logical flow diagram of subroutine TERP1	43
11	Logical flow diagram of subroutine TERP2	44
12	Logical flow of input data card sequencing	5 3

CONDOS--A MCDEL AND COMPUTER CODE TO ESTIMATE POPULATION AND INDIVIDUAL RADIATION DOSES TO MAN FROM THE DISTRIBUTION, USE, AND DISPOSAL OF CONSUMER PRODUCTS THAT CONTAIN RADIOACTIVE MATERIALS

F. R. O'Donnell, O. W. Burke, F. H. Clark, and L. R. McKay

ABSTRACT

A model and computer code (CONDOS) are described that estimate radiation doses to man from distribution, use, and disposal of a variety of consumer products that contain radioactive materials. CONDOS utilizes a generalized format in which the life span of a consumer product is divided into five main stages (distribution, transport, use, disposal, and emergencies) that require descriptions of the activities by which man may be exposed to the product (events) during each stage. These descriptions identify homogeneous groups of exposed persons and, thus, facilitate the selection of individuals who represent the exposed groups. The radiation doses associated with one year of product use to the total body and to selected reference organs and tissues of representative individuals can be estimated for each mode of exposure that is applicable to each event. Summation of the doses to representative individuals yields group and total population doses. An example of the use of CONDOS is given; the radiation doses associated with a hypothetical product are estimated for assumed conditions of exposure.

1. INTRODUCTION

A variety of consumer products containing radioactive materials are available to, or are being developed for use by, the general public.¹ These consumer products include a diversity of items, such as timepieces containing tritium or promethium-147, incandescent gas mantles containing thorium, and certain types of glazed ceramic tableware containing uranium. The distribution, use, and disposal of such products can present situations (events) in which man may receive radiation doses. Estimates of the dose equivalents^a that might be received by individuals, groups of individuals, and the entire population are required for evaluation of the radiological impacts associated with the consumer products.

A model and computer code (CONDOS) are being developed to provide a methodology for systematically estimating the radiation doses to man from such a diversity of consumer products. The model divides the life span of a consumer product into five main stages and provides a guide for description of the events. These descriptions identify homogeneous groups of exposed persons and, thus, facilitate the selection of individuals who represent the exposed population groups. The radiation doses associated with one year of product use to the total body and to the skin of the representative individuals can be estimated for each mode of exposure that is applicable to each event. Summation of the doses to representative individuals yields estimates of group and total population doses.

^aDose equivalent (rem) = absorbed dose (rad) X modifying factors. For the sake of convenience, "dose" is often interchanged with "dose equivalent" in this report.

A total dose has two components. One component results from radionuclides^b outside the body (external exposure), and the other component results from nuclides deposited in the body (internal exposure).^C External exposure is considered, according to the relevant exposure modes, in Section 3.1, and internal exposure is considered, according to the relevant modes of nuclide deposition, in Section 3.2.

The generalized model and the product information and mathematical techniques required to program the model are presented in Sections 2 and 3. A description of the FORTRAN -IV program CONDOS is presented in Section 4. The present version of the program provides estimates of the radiation doses, to the total body (from all radiations) and to the hands (from beta particles), associated with one year of distribution, use, and disposal of a consumer product containing natural uranium or thorium. The capability to provide dose estimates to other selected body organs and tissues will be provided in a future version of the program. Additional nuclides will be added to the data file so that a greater variety of products can be investigated.

^bFor the sake of convenience, "radionuclide" is shortened to "nuclide" in this report.

^C"Dose commitment" is frequently used to represent the total internal dose that will be received over a specified time period after intake of radioactive material (e.g., a 50-year dose commitment is the internal dose that will be delivered during the first 50 years after intake). This concept is used in this report. All internal doses are the 50-year dose commitments from intake of the radioactive material during the year being considered. Total doses to individuals and populations are the sums of the appropriate annual external doses and the appropriate 50-year dose commitments.

2. THE MODEL

The estimation of radiation doses to man from a consumer product requires (1) a description of the product; (2) a description of the life span of the product; (3) the identification and description of events, i.e., actions or acts by which man may be exposed to radiation from the product; (4) a determination of the conditions and modes of exposure relevant to each event; (5) a definition of all persons who may be exposed to radiation from the product; (6) the estimation of doses that may be received by each person exposed; and (7) an estimation of the total population dose attributable to the product.

The CONDOS model provides a guide for the implementation of items (1) through (5) and provides a mechanism for the achievement of (6) and (7). This model is described in the following sections.

2.1 Product Description

A description of the consumer product to be assessed, although not a part of the model <u>per se</u>, is required by the model. Such a description provides the basic information to be used in the estimation of doses and in the selection of parts of the product life span model that are applicable to the specific consumer product. The following information is required:

 names and quantities of the nuclides in the radioactive material which is in or on the product;

2. chemical and physical characteristics of the radioactive material, especially those that may be significant in the determination of release rates for the material, e.g., leach rate, vapor pressure, etc.;

3. density and thickness of the radioactive material;

4. density and thickness of the matrix (including the radioactive material) in which the radioactive material is incorporated;

5. density and thickness of any protective materials that may cover the radioactive material;

geometry of the product or of collections of the product,
 e.g., a shipping package;

7. identification of possible uses of the product;

8. identification of the means of transport, distribution, repair, and disposal of the product; and

9. identification of accident or misuse conditions to which the product might be subjected.

2.2 Model of Product Life Span and Conditions of Exposure

The model of the life span of a consumer product is outlined in Fig. 1, and the components that should be considered in completing the outline are listed in Table 1. The life span of a product is divided into five main "stages"--distribution, transport, use, disposal, and emergency or misuse situations. Although both distribution and transport are part of the physical transfer of the product from manufacturer to user, Table 1 shows that transport describes shipments of the product and distribution describes the presence of the product in distribution facilities such as warehouses.

Each stage is divided into a set of "substages" that delineate the general types of activities by which man could be exposed to the product.

ORNL-DWG 74-12377

Product Description	<u> </u>		Life	Span of a Consumer Pro	duct	
		1. Distribution	2. Transport	3. Use	4. Disposal	5. Emergencies
}		Substage 1.1	Substage 2.1	Substage 3.1	Substage 4.1	Accident from (1.1)
		Group 1.1.1	Group 2.1.1	Group 3.1.1	Group 4.1.1	Group A.(1.1).1
		Event 1.1.1.1	Event 2.1.1.1	Event 3.1.1.1	Event 4.1.1.1	Event A.(1.1).1.1
		Event 1.1.1.2				
	۲.	Event 1.1.1.NE	Event 2.1.1.NE	Event 3.1.1.NE	Event 4.1.1.NE	Event A.(1.1).NE
-		Group 1.1.2	Group 2.1.2	Group 3.1.2	Group 4.1.2	Group A.(1.1).2
		Event 1.1.2.1				
	T_>	Event 1.1.2.NE				
}	╾┰╼╞	Group 1.1.NG	Group 2.1.NG	Group 3.1.NG	Group 4.1.NG	Group A.(1.1).NG
Population		Event 1.1.NG.1	Event 2.1.NG.1	Event 3.1.NG.1	Event 4.1.NG.1	Event A.(1.1).NG.1
Information	۲	Event 1.1.NG.NE	Event 2.1.NG.NE	Event 3.1.NG.NE	Event 4.1.NG.NE	Event A.(1.1).NG.NE
		Substage 1.2	Substage 2.2	Substage 3.2	Substage 4.2	Accident from (1.2)
-	- 4 -	Group 1.2.1				
ب					11	
		Group 1.2.1G				
		Event 1.2.NG.1				
	` لـهـا`	Event 1.2.NG.NE				
դ		Substage 1.3				
		Substage 1.NS	Substage 2.NS	Substage 3.NS	Suistage 4.NS	Accident from (4.NS)
L.	-7-	Group 1.NS.1	Group 2.NS.1	Group 3.NS.1	Group 4.NS.1	Group A.(4.NS).1
``		Group 1.NS.NG	Group 2.NS.NG	Group 3.NS.NG	Group 4.NS.NG	Group A.(4.NS).NG
		Event 1.NS.NG.1	Event 2.NS.NG.1	Event 3.NS.NG.1	Event 4.NS.NG.1	Event A.(4.NS).NG.1
	۳ L	Event 1.NS.NG.NE	Event 2.NS.NG.NE	Event 3.NS.NG.NE	Event 4.NS.NG.NE	Event A.(4.NS).NG.NE

Figure 1. Outline of model of the life span of a consumer product.

Table 1. List of components to be considered in modeling the life span of a consumer product

Stages	Substages	Groups	Events (same for all groups)
Distribution	Factory	Sorters, packers, inspectors, handlers other workers, visitors, neighbors, passersby, any others	Each normal group activity, each accident or misuse
	Wholesale	See factory groups	
	Retail	Receivers, stock clerks, sales people, cashiers, other workers, examining cus- tomers, other customers, neighbors, pas- sersby, others	
	Representative	Representatives, customers, customers' neighbors, people along representatives' routes, others	
Tressport	Truck-long haul	Loaders, drivers, helpers, service per- sonnel, people at truck stops, people along routes, passing motorists, others	
	Truck-short haul	See truck-long haul	
	Truck-local delivery	See truck-long haul	
	Rail	Loaders, train crewmen, passengers, other rail workers, people in stations, people along railways, others	
	Air	Loaders, flight crew, passengers, other workers, people at airports, people along flight routes (accidents only), others	
	Bus	Drivers, passengers, depot workers, depot visitors, service personnel, people at bus stops, people along routes, passing motor- ists, others	

•

•

Table 1. List of components to be considered in modeling the life span of a consumer product (continued)

Stages	Substages	Groups	Events (same for all groups)
	Water	Loaders, crew, passengers, other dock workers, passing boats, people along waterways, people at docks, others	Each normal group activity, each accident or misuse
	Mail	Handlers, sorters, carriers, other postal workers, post office visitors, other substages for long distance mail, people along delivery routes	
Use	Use No. 1 Use No. 2 Etc.	Specify user groups whose methods of use are significantly different (e.g., use no. 1, 2,, frequent users, occa- sional users, etc.), people present during use, passersby, neighbors to users, others - similar for all uses	
	Maintenance	Maintenance-men, other workers, visitors, neighbors, others	
	Repair	Repairmen, other workers, visitors, neigh- bors, others	
Disposal	Collection	Collectors, drivers, unloaders, people along collection routes, passe osby , others	
	Incineration	Facility operators, nearby residents, dis- tant residents, people who frequent the area of the facility, passersby, others	
	Open-pit burning	See incineration	
	Conical burning	See incineration	
	Maceration	See incineration	

Table 1. List of components to be considered in modeling the life span of a consumer product (continued)

Substages	Groups	Events (same for all groups)
Compaction	See incineration	Each normal group activity,
Sanitary landfill	See incineration	each accident or misuse
Dump	See incineration	
Other methods	See incineration	
List each accident or misuse con- sidered in the preceeding groups	Firemen, policemen, medical teams, other emergency workers, nearby residents, passersby, other people involved in accident or misuse, spectators, others	
	Substages Compaction Sanitary landfill Dump Other methods List each accident or misuse con- sidered in the preceeding groups	SubstagesGroupsCompactionSee incinerationSanitary landfillSee incinerationDumpSee incinerationOther methodsSee incinerationList each accident or misuse con- sidered in the preceeding groupsFiremen, policemen, medical teams, other emergency workers, nearby residents, passersby, other people involved in accident or misuse, spectators, others

These activities are made more specific by subdivision into "groups" of functionally related activities.

Population information is supplied to the model at this point by determination of the number of persons engaged in the activities of each group and exposed to the product as a result of those activities. These persons are called group members.

The group activities are further divided into a set of actions during which group members may be exposed to the product--the "events". An event description contains the parameters required to solve the dose equations and defines an individual typical of the group exposed. This individual may be an average group member (one who represents the entire group), the maximally exposed member, or any other group member. The radiation doses estimated for each event are doses that the selected individual may receive from that event. Since the event descriptions prescribe the parameters to be used in the solutions of the dose equations, the correspondence of the events described with reality and how well they are described determine how well the estimated doses reflect actual doses.

The description of an event must include the following parameters, unless certain computer program options (Section 4.5) are selected:

 the probability that a member of the group will be involved in the event,

2. the time periods during which the individual will experience each mode of exposure,

3. the amount of radioactive material in the product.

4. the number of point sources chosen to represent the source,

5. the Cartesian coordinates of each point source and of the receptor,

6. the attenuation (shielding) factor for the composite of all materials between each point source and the receptor,

7. the amount of radioactive material ingested,

8. the concentration of radioactive material in air,

9. the effective radius of the room in which the material is airborne,

10. the rate of leakage of material from the product to the room or the total amount of material that has leaked into the room.

11. the air turnover rate for the room, and

12. the volume of the room.

Some of the substages and groups listed in Table 1 may not be required to model the life span of a particular consumer product. The modeler is responsible for choosing the substages, groups, and events that are appropriate to the description of the life span and for deleting those that are not appropriate.

All event descriptions must be supplied by the user of this version of CONDOS. The inclusion of a set of standard events which the user may select is being considered for a future version. Descriptions of the normal activities of the members of each group may require specification of more than one event because some of the activities may present very different conditions of exposure. Similarly, more than one accident or misuse event may be required.

2.3 Methodology for Estimation of Radiation Doses to Man

The methodology used in CONDOS for the estimation of potential radiation doses to man from a consumer product containing radioactive

material consists of three parts: information assembly, life span model construction, and dose estimation. This section shows how the three parts are combined to form the methodology. Figure 2 illustrates the flow of operations used in the methodology.

Product information (Section 2.1) and population information (the number of persons potentially exposed to the product and their behavioral patterns in exposure situations) are gathered and sorted. This information is used to construct the life span model for the consumer product.

The model is constructed, as shown in Figure 2, by division of each of the five stages (1. - 5.) into an appropriate set of substages, e.g., stage "3. Use" is divided into substages $3.1, \ldots, 3.X, \ldots, 3.NS$. Each substage is then divided into a set of groups, e.g., substage 3.X is divided into groups $3.X.1, \ldots, 3.X.Y, \ldots, 3.X.NG$. The number of persons (NM) in each group^a is determined at this point. The various activities by which the members of each group may be exposed to the product (the events) are identified next for each group, e.g., group 3.X.Y is involved in events $3.X.Y.1, \ldots, 3.X.Y.2, \ldots, 3.X.Y.NE$.

The model is completed by selection of an individual from the group who partakes in one event, description of the event, and repetition of the selection and description process for every event identified for every group. The individual selected determines how the event should be described and, thus, determines the conditions of exposure that will

^dNM may be taken as the total number of group members (e.g., all countermen who work in diners) or may be restricted to the number of group members potentially exposed to the product (e.g., all countermen who work in diners that use the product). Selection of the restricted number is preferred.



Figure 2. Flow diagram of the methodology for estimation of radiation doses to man from a consumer product containing radioactive material.

be used to quantify the parameters required to solve the dose equations. If an average individual (one who represents the entire group) is selected, average conditions of exposure are chosen in the event description. If the maximally exposed individual is selected, conditions of maximum exposure are selected. The selection of individuals should be consistent for all events and for all groups, i.e., only average individuals should be selected, or only maximally exposed individuals should be selected.

Each event description is input to the computer code (Section 4). The conditions of exposure for each event, e.g., event 3.X.Y.Z in Figure 2, are used to calculate potential radiation doses to the total body (Section 3) of each selected individual by four exposure pathways: direct exposure to gamma rays (DIR) emitted from the radioactive material in or on the product, immersion in air contaminated with radioactive material (EM) released from the product, and ingestion (ENG) and inhalation (EN) of released material. The total dose to the individual (TDI) is the sum of the doses delivered via the four exposure pathways. Each TDI contributes to the collective dose received by all group members (the group dose, TGD). The contribution of each TDI is called an event dose, e.g., ED(3.X.Y.Z), and is quantified by weighting TDI by NM and by the joint probability (EVPROB)^a that a group member will partake in the event and that the event will occur. An ED is calculated for each event.

^a If an average individual participates in a common event (one that will occur every time the group is exposed to the product), EVPROB = 1. If the maximally exposed individual is involved, EVPROB = 1/NM. For other individuals $1 \ge EVPROB \ge 1/NM$. If the event is not a common one, its probability of occurrence will be <1, e.g., = P. In this case, for the individuals noted above, EVPROB = P, P/NM, or in general, P \ge EVPROB \ge P/NM, respectively.

Population doses, the sum of the total body doses to each member of a population group, are estimated for each group, substage, stage, and the entire population as follows. Group doses are the sum of the appropriate event doses, e.g., TGD(3.X.Y.) = $\sum_{\substack{\Sigma\\7-3}}^{NE}$ ED(3.X.Y.Z). A TGD is

calculated for all the groups that form a substage, e.g., 3.X.1,..., 3.X.NG, and the substage dose is estimated by summation of the pertinent group doses, e.g., TSSD(3.X) = $\frac{NG}{\Sigma}$ TGD(3.X.Y). A TSSD is esti-Y=1

mated for all substages delineated in the model. The potential collective dose to all persons involved in a stage of the product life span is obtained by summation of the appropriate substage doses, e.g., TSD(3) = NS Σ TSSD(3.X). A similar sequence of calculations is made to obtain X=1 the remaining TSDs, and the total population dose associated with one year of product availability is estimated by TPD = $\frac{5}{\Sigma}$ TSD(S).

CONDOS also provides an estimate of the radiation dose to the skin from contact with a product containing radioactive material that emits beta particles. The present version of the computer code estimates this beta dose only for a single event, and provides the estimated dose for a specified period of contact (Section 3.1.3). The computer code is listed in Appendix D.

2.4 Discussion

CONDOS provides a methodology for estimating potential radiation doses to man from consumer products containing radioactive materials. These doses can, in principle, be estimated fairly accurately. That is, the dose equations (Section 3) will calculate the doses fairly accurately if the input data from the "real-world" descriptions of the product and its life span are accurate. Thus, the accuracy of the dose estimates is limited essentially by the accuracy of the "real-world" descriptions.

A product can be well defined, but such a definition will frequently be unavailable. A manufacturer may be unable to supply the type of information needed (He might not have it). For example, a product might be described as tableware coated with a ceramic glaze containing 14 wt% uranium. The desired description would specify the numbers, sizes, and shapes of the various forms of tableware, the composition, and thickness of the glaze, and the identities and quantities of the nuclides present. In addition, several manufacturers of the product could exist, and each might manufacture a product with a design that is different from the designs of all other manufacturers.

The life span of a product cannot be defined precisely. Each of the infinitely large number of possible events that can occur during distribution, use, and disposal of a product cannot be considered individually. An average, or typical, event that represents many closely related events must be constructed, if the modeler is not to be overwhelmed by voluminous amounts of data. Such a construction must introduce uncertainties in the data. For example, a person might spend 8 hr/day between 1 and 100 m from a source with various absorbers between the person and the source. An accurate description, which might require 100 events, would specify the time, distance, and absorber for each different event. Construction of the average event might describe an exposure of 8 hr/day at 50 m with an average shielding factor to account

for the absorbing material. If the relative movements of products and individuals and the interposition of several, mostly uncharacterized, absorbers during these movements are considered, the complexity of life span definition becomes apparent.

Because of the uncertainties and variations possible in product definition and life-span modeling, some input parameters used to solve the dose equations could be in error by as much as a factor of 10. Careful selection of the parameters could reduce the errors appreciably.

The expected errors in input data were considered in and have influenced the models of the dose equations. Some of the calculations and nuclear data files were simplified by procedures that might introduce a combined error of less than \pm 10% into the dose estimates. This error is small compared with expected input data errors. The sources of the errors are discussed in other sections of this report.

Another important factor is that the present version of CONDOS is designed primarily to estimate doses from products containing natural uranium or thorium. Therefore, the dose equation models and computer program logic are designed to accommodate heavy-element decay chains, not individual nuclides. However, this does not restrict the utility of CONDOS, because only minor modifications in certain program functions are required to add such a capability.

3. MATHEMATICAL MODELS OF THE DOSE CALCULATIONS

The potential doses from a consumer product to representative individuals and population groups that are considered by CONDOS are discussed in Section 2.3. The mathematical equations and their models,

as they appear in the computer code, are presented here. Appendix D is a listing of the computer code.

3.1 External Exposure

The modes of external exposure considered in CONDOS are direct irradiation, immersion in a contaminated cloud, and contact with a beta-particle-emitting surface.

3.1.1 Dose from direct irradiation

The direct radiation dose, in an absorbing medium, from gama photons emitted from a consumer product represented by an array of point sources is computed by the equation:²

$$DE = \frac{f \cdot (MF)}{4\pi} \sum_{k=1}^{NP} \sum_{i=1}^{NG} E_{ik} C_{ik} B_{ik} \left(\frac{S_{ik}}{r_k^2}\right) \exp(-\mu_{ik} r_k)$$

where

DE = dose equivalent rate (rem/hr), i = photon index, k = source index, NG = number of photons emitted by k th source, NP = number of point sources, f = dose conversion factor (rad/r), MF = modifying factor (rem/rad), E_{ik} = energy (MeV/photon) of i th photon emitted by k th source, C_{ik} = exposure rate-energy flux ratio (r-hr⁻¹/MeV-cm⁻²-sec⁻¹) for photon energy E_{ik},

- B_{ik} = dose build-up factor (dimensionless) for i <u>th</u> photon from k <u>th</u> source in the chosen absorbing medium, a function of E_{ik} and $\mu_{ik}r_k$,
- Sik = emission rate (photons/sec) of the i th photon from the k th
 point source,
- μ_{ik} = linear attenuation coefficient (cm⁻¹) for i <u>th</u> photon from k th source in chosen absorbing medium,
 - r_{ν} = distance (cm) from k <u>th</u> source to receptor.

A typical consumer product will contain several nuclides that are part of a decay chain. Therefore, a simple counting of photons of various energies becomes complicated. A general method for counting photons is adopted in CONDOS. The gamma ray energy spectrum is divided into 25 energy intervals (see Appendix A) and the photons emitted by each nuclide are placed in the appropriate energy interval. In addition, all energydependent quantities are defined in terms of these energy intervals and those that are stored as data variables in the code are listed in Appendix A.

Since the source term for a consumer product will probably be given as the number of grams of radioactive material contained, a method for relating the amount of each nuclide present in the material is provided. Since only natural uranium and thorium are of present interest, the factors necessary to convert from grams of material to photons of a specific energy/sec are provided in the code as data variables. These variables are also listed in Appendix A. The model of the dose equation for direct irradiation is:

DIR = 79.6*TDIR*
$$\sum_{I=1}^{NGINT} ENRGY(I)*CONVR(I)* \sum_{J=1}^{NN} ACT(J)*FGM(I,J)*$$

NP
 $\sum_{K=1}^{NP} S(K)*ATN(I,K)*AATN(K)*R(K)^{-2}$ (1)

where

DIR = direct dose equivalent rate (mrem/year),

- I = energy interval index,
- J = nuclide index,
- K = source index,
- NGINT = number of photon energy intervals,
 - NN = number of nuclides in the product,
 - NP = number of point sources,
 - 79.6 = dimensional conversion factor (1000 mrem-rem $^{-1}/4II$),
 - TDIR = duration (hr/year) of exposure,
 - IEL = radioactive material index (1 → thorium;

 $2 \rightarrow \text{uranium}$),

- ENRGY(I) = average energy (MeV/photon) of a photon in the I \underline{th} energy interval,
- CONVR(I) = exposure rate-energy flux ratio for I <u>th</u> energy photons,(r-hr⁻¹/MeV-cm⁻²-sec⁻¹)

$$ACI(J) = YI(J)*RA(J)*CH(J)*AVO*DEC(IEL)/3600$$

= disintegration rate (dps/g) of J th nuclide per gram of
IEL th material,

- YI(J) = abundance (g/g) of the mother nuclide of the decay chain containing the J <u>th</u> nuclide in the material,
- RA(J) = ratio (dps/dps) of the disintegration rate of the J <u>th</u> nuclide to that of its mother,

$$CH(J) = \frac{\lambda m/Mm}{DEC(IEL)}, \text{ ratio (mole-g}^{-1} \text{ hr}^{-1}/\text{mole-g}^{-1} - \text{hr}^{-1}) \text{ relating the disintegration rate per gram of the mother nuclide (m)}^{a}$$
of the decay chain containing the J th nuclide to that of the primary nuclide (p),^b

$$\lambda_{\chi}$$
 = decay constant (hr⁻¹) for the x th nuclide,

$$M_{\chi} = molecular weight (g/mole) of the x th nuclide,$$

DEC(IEL) =
$$\lambda_p/M_p$$
, constant (mole/g-hr) for the IEL th material that is
proportional to the disintegration rate^a of one gram of the
primary nuclide^b,

- FGM(I,J) = number of photons (photon/dis) with energy in the I <u>th</u> interval emitted per disintegration of the J <u>th</u> nuclide,
 - S(k) = amount (g) of the IEL <u>th</u> material in or on the k <u>th</u> source,

 $^{^{}a}\text{AVO}$ * λ_{x}/M_{x} = disintegration rate per gram (dis/hr-g) of the x $\underline{\text{th}}$ nuclide.

^bThe primary nuclides are chosen to be U-238 in natural uranium and Th-232 in natural thorium.

 $ATN(I,K)^{C}$ = shielding factor (dimensionless) for k th source and photons in I th energy group,

AATN(K)^C = shielding factor (dimensionless) for k th source-taken the same for all photon energies,

$$R(K) = \{[X(K) - X0]^2 + [Y(K) - Y0]^2 + [Z(K) - Z0]^2\}^{1/2}$$

- = distance (cm) between the k th source and the receptor.
- X(K), Y(K), Z(K) = coordinates (cm) of the k th source

X0,Y0,Z0 = coordinates (cm) of the receptor.

Use of this equation requires that, for any exposure situation, the source be described as a single point source or as a distribution of such sources. The distance R(K) between each source and the receptor must be specified. This may be done in three ways:

- set the receptor at the origin (X0=Y0=Z0=0) and specify the desired distance as X(K); set Y(K)=Z(K)=0, or
- (2) set the receptor at the origin and specify each point source coordinate, or

(3) specify the coordinates of the receptor and each point source.

Because the information required to calculate explicit attenuation factors will probably be unavailable, only the option to specify gross shielding factors is included. These factors may be energy independent $[0 \le AATN(K) \le 1]$ or dependent $[0 \le ATN(I,K) \le 1]$ and must be specified for each point source. The latter option will facilitate the eventual incorporation

^CFor a given point source (K), either AATN(K) or ATN(I,K) must be set equal to 1 (see Section 4.6), i.e., one variable is not used in the calculation. If AATN(K) is used, its value is an input variable. If ATN(I,K) is used, it must be evaluated for each of the 25 values of 1.

of the capability to deal directly with volume source distributions, the most general case. The option to calculate explicit attenuation factors can be added, if found to be necessary.

The dose conversion factor (rad/r) and modifying factor (rem/rad)were set equal to unity and not stated explicitly in the model equation. These approximations should not introduce significant errors into the results unless the dose to bone is of importance.² The backscatter correction was also set to unity; a procedure which could lead to small underestimates of the direct dose in some exposure situations of interest.

3.1.2 Dose from immersion in contaminated air

The dose equivalent rate (EH) from photons emitted from radioactive material suspended in air is calculated by the equation:

where

```
EM = immersion dose equivalent rate (mrem/year),
J, NN and ACT(J) are defined in Eq. (1):
(3.74E4*8760) = dimensions conversion factor (dps/µCi)*(hr/year),
CONC = concentration (g/cm<sup>3</sup>) of radioactive material in air,
TIMM = duration (hr/year) of immersion in contaminated air,
F(ABS) = correction factor (dimensionless) to compute dose in a medium
of finite extent relative to that in an infinite medium,
CIMM(J) = dose conversion factor (mrem-year<sup>-1</sup>/µCi-cm<sup>-3</sup>) for
immersion in air contaminated with the J th suclide.
```

The source term CONC may be expressed by one of three methods (see Section 4.2.2): (1) as a specified concentration of radioactive material containing the J th nuclides, (2) as a calculated steady-state concentration from a specified leakage rate of radioactive material into a finite air space (e.g., a room) that has a specified air exchange rate, or (3) as a calculated concentration from a specified amount of material which has leaked into a specified volume (a room) with poor ventilation (or no air exchange).

The correction factor, F(ABS), can be quite important if the air volume is defined by a closed room. F(ABS) is provided by a FUNCTION subroutine and requires input for the effective radius of the room (see Section 4.2.1).

The EXREM III³ computer code which calculates the dose rate in an infinite volume of contaminated air was used to provide the dose conversion factor for each nuclide. Each nuclide that contributes significantly to the dose from 20-year-old natural uranium and thorium (see Appendix A) was processed. These factors, CIMM(J), are stored as data variables in the program and are listed in Appendix A.

3.1.3 Dose to skin from contact with a beta-particle source

The beta dose to the skin can be important if a product containing beta-emitting nuclides is handled. The beta dose equation is derived from the well-known Loevinger point-source dose-rate formula.^{4,5} Details of the derivation are in Appendix B. Dose-rate equations (B.4) and (B.5) are represented by:

$$DBETA = PMEG*TBETA*(SOB/ROB)*AVO*DEC(IEL)* \Sigma EA(I)*THETA(EO(I))*$$

$$I=1$$

BCOR(STH, HTH, EO(I), IAR)*BCON(I)*
$$\Sigma$$
 FBT(I,J)*CC(J) (3)
J=1

where

PMEG = dimensional conversion factor $(1/2 \times 1.6E-8 \text{ g-rad/MeV})$,

ROB = density
$$(g/cm^3)$$
 of the matrix material plus the radio-
active material imbedded in it,

AVO, DEC(IEL), J, and NN were defined in equation (1),

I = beta-particle energy group index (dimensionless),

NBINT = number (dimensionless) of beta-particle energy groups,

EA(I) = mean energy (MeV/beta) of a distribution of beta particles whose maximum energy falls within the I <u>th</u> energy group,

THETA(EO(I)) = Θ^{-1} , see Appendix B,

EO(I) = maximum energy (MeV/beta) assigned to a distribution of beta particles whose actual maximum energy lies within the boundaries of the I th energy group,

BCOR(STH,HTH,EO(I),IAR) = { } in equation (B.4) if IAR = 1, or in equation (B.5) if IAR = 2, see Appendix B,

BCON(I) = dose conversion factor (rem/rad), STH = absorbing layer thickness (mg/cm²), see Appendix B, HTH = beta source thickness (mg/cm²), see Appendix B,

- FBT(I,J) = number of beta particles with energies in the I <u>th</u>group emixted per disintegration of the J <u>th</u> nuclide(beta/dis),
 - CC(J) = YI(J)*RA(J)*CH(J), all of which are defined in equation (1).

The solution of equation (3) is expressed in the output table as the absorbed dose (rad) from TBETA hours of contact with the product. This form of expression disregards the dose conversion factor BCON(I), which has been set equal to unity for all I in the computer code, thus allowing DBETA to be expressed as an absorbed dose rather than as a dose equivalent.

The user must supply SOB, ROB, STH, HTH, and TBETA. If more convenient, the weight fraction [g(IEL)/g(matrix)] of the radioactive material in the matrix may be substituted for SOB and ROB set equal to unity.

3.2 Internal Exposure

The modes of internal exposure considered in CONDOS are ingestion and inhalation of a radioactive material. All types of radiations are considered to contribute to the dose.

3.2.1 Dose from ingestion

The 50-year dose commitment to the total body of an individual who ingests radioactive material during the year being considered is given approximately by: ENG = $(AMT/37)* \sum_{J=1}^{NN} ACT(J)*CING(J)$ (4) where

ENG = 50-year dose commitment (mrem/year) to the total body from ingestion during the year being considered,

J, ACT(J), and NN are defined in equation (1),

AMT = amount (g/year) of radioactive material ingested,

37 = dimensional conversion factor (3.7E4 dps/µCi)/(1000 mrem/rem),

CING(J) = 50-year dose-conversion factor (rem/µCi) to the total body from ingestion.

The INREM⁶ computer code which calculates internal dose commitments from intake of radioactive materials was used to determine the values of CING(J). The CING(J) for 50-year, total-body dose commitments from ingestion of 1 μ Ci of the nuclides of interest, are stored as data variables in the program and are listed in Appendix A. The user must supply only AMT to obtain ENG. The dose commitments to other organs can be calculated by equation (4) if the appropriate dose-conversion factors [CING(J)] are used.

3.2.2 Dose from inhalation

The 50-year dose commitment to the total body of an individual who inhales radioactive material during the year being considered is given approximately by:

$$EN = CONC*TINH*(BR/37)* \sum_{J=1}^{NN} ACT(J)*CINH(J)$$
(5)

where

EN = 50-year dose commitment (mrem/year) to the total body from inhalation during the year being considered,
J, ACT(J), and NN are defined in equation (1),

CONC = concentration (g/cm^3) of radioactive material in air, see sections 3.1.2 and 4.2,

TINH = time (hr/year) spent breathing contaminated air,

BR = breathing rate $(0.833E+6 \text{ cm}^3/\text{hr})$ of average person,

37 = dimensional conversion factor (3.7E4 dps/µCi)/(1000 mrem/rem),

CINH(J) = 50-year dose-conversion factor (rem/µCi) to the total body from inhalation.

The INREM⁵ code was used also to obtain the values of CINH(J). The values for 50-year, total-body dose commitments from inhalation of 1 μ Ci of the nuclides of interest are stored as data variables and listed in Appendix A.

The user must provide TINH and the means to compute CONC as discussed in section 4.2. Equation 5 can calculate dose commitments to other body organs, if the appropriate dose-conversion factors are used.

4. THE COMPUTER CODE

Two versions of the CONDOS computer code exist. Both versions are written in FORTRAN IV; one for use in the batch mode on the IBM 360 75/91 computer and one for use in the time-shared mode on the DEC PDP-10 computer. The only differences between the two versions are in the small number of details that make them compatible with the respective computers.

4.1 The Computer Solution

The computer code solves equations (1), (2), (4), and (5) and provides the logic required to read-in the input parameters supplied

by the event descriptions given in the model, to implement the calculational sequence prescribed by the methodology, and to output the results of the calculations in tabular form. This subsection describes what the code does; the remaining subsections show how it is done.

Doses to the total body of selected individuals are calculated both for external exposure pathways (direct irradiation and immersion) and for internal exposure pathways (ingestion and inhalation). The doses from external exposure are annual dose equivalents (mrem/year -interpreted as the dose from one year of product availability), but those from internal exposure are the integrated dose equivalents that will be received during the first 50 years after intake of the radioactive material (50-year dose commitments). The internal doses are expressed also in mrem/year and are to be interpreted in the same context as the external doses, i.e., as doses from one year of product availability.

The computer code lists the potential doses to each individual considered in the model according to exposure pathway. The doses to each individual are summed and the resultant total individual doses are also listed in the output table. Appendix C contains a sample output table.

Potential doses to population groups involved in the life span of the product are computed by summation of the total individual doses, according to the sequence prescribed by the methodology (Section 2.3). These population doses (group dose, substage dose, stage dose, and total population dose) are listed also in the output table. These doses are expressed in man-rem/year and are interpreted as the collective dose from one year of product availability to all individuals in the group being considered.

The code also solves equation (3) and provides a statement of the beta dose to the skin (rads) for a specified period of contact with the product.

4.2 Auxiliary Calculations

4.2.1 Finite volume correction factor

The exposure rate, neglecting scattering, at the origin (center) of a spherical volume of contaminated air is

ER =
$$\int_{0}^{2} \frac{F_1 S_v \exp(-\mu r)}{4\pi r^2} 4\pi r^2 dr$$
,

where

ER = exposure rate (r/hr), F_1 = dimensional conversion factor (r-hr⁻¹/photon-sec⁻¹-cm⁻²), S_v = source strength (photons/sec-cm³) per unit volume R = radius (cm) of volume,

 μ = linear absorption coefficient (cm⁻¹).

The solution is, $ER = (F_1 S_v / \mu) [1 - exp(-\mu R)]$. For an infinite volume,

 $ER = F_1 S_v / \mu.$

Therefore, the ratio of the dose rate from a volume of radius R to that from an infinite volume is

 $F(ABS) = 1 - exp(-\mu R).$

This correction factor was used in equation (2) and is calculated by the FUNCTION F(ABS) subroutine. Since μ is dependent on the photon

•

2

ľ

energy, a characteristic value of μ is provided by the code for each nuclide. The subroutines TERP1 and TERP2 provide the energy-dependent values of μ .

4.2.2 Concentration of radioactive material in air

The variable CONC, the concentration of radioactive material in air, which is used in equations (2) and (5), can be specified in one of three ways. The index ICON selects the method for specification of CONC.

1. ICON = 1. The user supplies the air concentration in g/cm^3 .

2. ICON = 2. This option calculates the equilibrium air concentration in a ventilated room with an air exchange rate, VRATE (cm^3/hr), and inleakage of radioactive material, SRATE (g/hr). If the room has a volume V (cm^3), the time-dependent concentration of material in the room is

$$\frac{d CONC}{dt} = \frac{SRATE}{V} - \frac{VRATE}{V} * CONC$$

At equilibrium, $\frac{dCONC}{dt} = 0$, and CONC = SRATE/VRATE .

SRATE AND VRATE must be specified by the user, if this option is chosen.

3. ICON = 3. The air concentration in an unventilated room is computed. In this case, CONC = TLEAK/V, where

TLEAK = the total amount (g) of material leaked into the room, $V = volume (cm^3)$ of the room.

4.2.3 Geometric conversions

The dose from a plane source may be calculated as follows. Let G(r) be the point source kernel for an isotropic point source of unit strength that is imbedded in an infinite homogeneous medium. If a disk of radius RD (Fig. 3) and source strength SD per unit area is imbedded in such a medium and is represented by an array of point sources, the dose rate at point P, normal to the center of the disk, is

$$DR(P) = \int_{0}^{RD} G(r) (2\pi SD \rho d\rho) = 2\pi SD \int_{x}^{r} G(r) dr, \text{ since}$$

$$r^{2} = x^{2} + \rho^{2}.$$

If the plane is of infinite extent,

$$DR(P) = 2\pi SD \int_{X}^{\infty} r G(r) dr .$$

Replacement of a distributed volume source by an approximately equivalent surface source is sometimes desirable. This can be accomplished by the approximation

$$SS = SV \star T$$
 ,

where

SS = equivalent surface source, SV = volume source,





Figure 3. Coordinate system for conversion from point to disk sources.

T = {thickness of volume source if T < $1/\mu$, { $1/\mu$ otherwise,

 $1/\mu$ = mean free path (cm) of the radiation in the absorbing medium.

4.3 Logical Flow of the Code

The main program controls the flow of operations performed by the computer code (Fig. 4). In addition, it reads in the data contained in the bulk storage area. Appendix D contains a complete program listing.

After initialization of the variables, the program calls subroutine INPUT to read, place in disk storage, and write out the input data. Subroutine PREFAS is then used to compute the beta hand dose. The format of the output table is generated by use of subroutine OUTPUT.

The remaining dose calculations are performed by subroutine CALC after the appropriate input variables are read from the disk storage area. CALC provides the doses to representative individuals and to the total group. These doses are recorded in the output table by subroutine OUTPUT.

The dose calculations are repeated and printed out for each event in which each group, substage, and stage is involved. Group, substage, and stage population doses are summed to provide, respectively, the substage, stage, and total population doses. These doses are printed in the output table. Finally, the beta hand dose is printed out.

4.4 Description of Subroutines

Several calculations are performed by subroutines. The main program calls directly on subroutines INPUT, PREFAS, OUTPUT, and CALC.



Figure 4. Logical flow diagram of the main program.

Subroutine PREFAS uses the function subroutines FC(EO), FNU(EO), BCOR(IR,EO, S,H). and THETA(EO). Subroutine CALC uses the function subroutine F(ABS) which itself calls subroutines TERP1 or TERP2.

<u>Subroutine INPUT</u> reads the input data from the input device, stores it on a bulk storage device (e.g., disk), and causes it to be written by an output device. Figure 5 shows the logical flow of INPUT.

<u>Subroutine PREFAS</u> provides the solution to equation (3), the beta dose to the hands from contact. In addition, some of the variables used in the other dose calculations are computed. These include AVO, GAMCO, CC(J), and FBET(J) = FGM(I,J)*CC(J)*ENRGY(J)*CONVR(I), all of which are defined in equation (1). The logical flow of PREFAS is given in Fig. 6.

Function subroutines FC(EO) and FNU(EO) supplies the values of c and v, respectively, for use in specifying the form of the Loevinger function used in equation (3) and in solving that equation.

<u>Function subroutine THETA(EO)</u> specifies the values of THETA to be used in the solution of equation (3).

<u>Function subroutine BCOR(IR, EO, S, H</u>) computes the values of the Loevinger function for equation (3). Solutions to both cases specified by equations (B.4) and (B.5) are available.

<u>Subroutine OUTPUT</u> writes out the descriptive information supplied for the problem and generates the format of the output table. The doses computed by subroutine CALC are printed in the output table and a running summation is kept of their contributions to the total group, substage, stage, and population doses. These sums are printed in the table after each group, substage, stage, and all stages have been considered. Figure 7 shows the logical flow of OUTPUT.



Figure 5. Logical flow diagram of subroutine INPUT.



ORNL-DWG 74-12382

Figure 6. Logical flow of subroutine PREFAS.



Figure 7. Logical flow of subroutine OUTPUT.

<u>Subroutine CALC</u> solves equations (1), (2), (4), and (5) for every event that "Ecified in the life span model of a consumer product. Summation of these doses for each event provides the representative individual's total dose. Multiplication by the number of group members and the probability that a member will be involved in the event provides the contribution of that event to the total dose associated with the group. Summation of these contributions yields the total group dose. Figure 8 gives the logic of subroutine CALC.

<u>Function subroutine F(ABS</u>) follows the logic given by Fig. 9. It calculates the finite volume correction factor given in section 4.2.1.

Subroutines TERP1 and TERP2 provide the linear absorption coefficient, μ , discussed in section 4.2.1 and used in calculating F(ABS). Both select the appropriate value for μ from stored data tables. TERP1 is used if the photon energies occur in equal increments; TERP2, if they do not. The logical flow of the subroutines are given in Figs. 10 and 11, respectively.

4.5 Definition of Program Variables

The FORTRAN variables used in CONDOS are defined in this section. These variables are listed in alphabetical order in three groups: the input variables, the more important internal variables, and the output variables. The subscript I denotes the I <u>th</u> energy group for photons or beta particles; the subscript J denotes the J <u>th</u> radionuclide; the subscript K denotes the K <u>th</u> point source; and the subscript IEL denotes natural thorium (IEL = 1) or natural uranium (IEL = 2). Variables marked A are obtained from the data base; variables marked B are problem input.



Figure 8. Logical flow of subroutine CALC.



Figure 9. Logical flow of function subroutine F(ABS).





Figure 10. Logical flow of subroutine TERP1.



Figure 11. Logical flow of subroutine TERP2.

.

Input variables

В	AMT	Amount (g/year) of radioactive material ingested
В	AATN(K)	Energy-independent photon attenuation coefficient
		(dimensionless); see section 3
В	ATN(I,K)	Energy-dependent photon attenuation coefficient
		(dimensionless); see section 3
A	AVO	Avogadro's number (0.6023E24 atoms/mole)
Α	BCON(I)	Dose-conversion factor (rem/rad) for beta particles
Α	BR	Average breathing rate (0.833E6 cm ³ /hr of an adult
A	CH(J)	Ratio (dimensionless) relating the disintegration rate
		per gram of the mother nuclide of the decay chain con-
		taining the J <u>th</u> nuclide to that of the primary nuclide;
		see section 3.1.1
A	CIMM(J)	Dose-conversion (mrem-year $^{-1}/\mu$ Ci-cm $^{-3}$) for immersion;
		see section 3.1.2
A	CING(J)	Dose-conversion factor (rem/ μ Ci) for ingestion;
		see section 3.2.1
Α	CINH(J)	Dose-conversion factor (rem/ μ Ci) for inhalation
		see section 3.2.2
В	CONC	Concentration (g/cm ³) of radioactive material in air;
		see sections 4.2.2 and 3.1.2
А	CONVR(I)	Exposure rate energy flux ratio (r-hr ⁻¹ /MeV-cm ⁻² -sec ⁻¹)
		for direct irradiation by photons; see section 3.1.1
A	DEC(IEL)	Constant (mole/g-hr) for the IEL <u>th</u> radioactive mate-
		rial that is proportional to the disintegration rate
		of one gram of the primary nuclide, see section 3.1.1

- A DENRG Energy increment (MeV) between data points in TABMU; see TERP subroutines
- A DENRG1 Energy increment (MeV) between data points in TABMU1; see TERP subroutines
- A EA(I) Mean beta particle energy (MeV/beta) in the I <u>th</u> energy group; see section 3.1.3
- A ENRG(J) Photon energy (MeV) representative of photons emitted by a nuclide; see TERP subroutines
- A ENRGH Maximum photon energy (MeV) considered in TABMU
- A ENRGH1 Maximum photon energy (MeV) considered in TABMU1
- A ENRGL Minimum photon energy (MeV) considered in TABMU
- A ENRGL1 Minimum photon energy (MeV) considered in TABMU1
- A ENRGY(I) Energy of the average photon (MeV/photon) in the I <u>th</u> energy group
- A EO(I) Maximum beta particle energy (MeV/beta) in the I <u>th</u> energy group; see section 3.1.3
- B EVPROB The joint probability (dimensionless) that a group member will be involved in a particular event and that the event will occur, see section 2.3
- A FBT(I,J) Number of beta particles with energies in the I \underline{th} group emitted per disintegration of the J \underline{th} nuclide (beta/dis); see section 3.1.3
- A FGM(I,J) Number of photons with energy in the I <u>th</u> group emitted per disintegration of the J <u>th</u> nuclide (photon/dis); see section 3.1.1
- B GROUP Name (alphanumeric) of stage being considered; see sections 2.2 and 4.6

В	HTH	Thickness (mg/cm ²) of beta particle emitting source
В	ICON	Flag variable (dimensionless) for selection of the
		method for calculating CONC; see section 4.2.2
В	IGP	Stage identification index (dimensionless)
A	INCR	Index variable (dimensionless) for selection of
		TERP1 or TERP2 in subroutine F(ABS)
В	ISP	Substage identification index (dimensionless)
В	ITN	Index variable (dimensionless) for execution of
		option to select ATN(I,K) or AATN(K)
В	IVENT	Name (alphanumeric) of event being considered; see
		sections 2.2 and 4.6
А	JHI(IEL)	Index (dimensionless) corresponding to J for specifi-
		cation of the last of a sequence of nuclides in the
		materia]
А	JLO(IEL)	Index (dimensionless) for specification of the first
		nuclide in a sequence of nuclides in the material
A	NBINT	Number (dimensionless) of beta particle energy groups
A	NGINT	Number (dimensionless) of photon energy groups
В	NM	Number of group members (persons); see section 2.2
В	NOEV	Number (dimensionless) of events to be considered for
		the group being studied; see section 2.2
В	NP	Number (dimensionless) of point sources used to repre-
		sent the product
В	R	Effective diameter (cm) of contaminated air space
А	RA(J)	Ratio (dimensionless) of the disintegration rate of
		the J \underline{th} daughter nuclide to that of the mother of
		its chain

48

В	ROB	Density (g/cm ³) of matrix material which contains
		beta-emitting nuclides and a binder; see section 3.1.3
В	S(K)	Amount (g) of radioactive material in or on K <u>th</u>
		source
В	SOB	Concentration (g/cm 3) of beta-emitting material in
		matrix material
В	SRATE	Rate (g/sec) of radioactive material leakage from the
	•	product
В	SSGRP	Name (alphanumeric) of group being considered; see
		sections 2.2 and 4.6
В	STH	Thickness (mg/cm ²) of absorbing layer between beta-
		particle-emitting source and dose point, see section
		3.1.3
В	SUBGRP	Name (alphanumeric) of the substage being considered;
		see sections 2.2 and 4.6
А	TABMU	Table of energy-dependent linear absorption coefficients
		(cm ⁻¹) for photons of energy greater than 0.2 MeV
А	TABMU 1	Table of energy-dependent linear absorption coefficients
		(cm ⁻¹) for photons of energy less than 0.2 MeV
В	TBETA	Duration (hr) of direct contact between skin and beta
		particle-emitting source
В	TDIR	Duration (hr/year) of exposure to photons emitted
		from an external source
В	TIMM	Duration (hr/year) of immersion in a volume of air
		contaminated with photon-emitting nuclides
В	TINH	Duration (hr/year) of inhalation of contaminated air

В	TITLEI	Two lines of descriptive information, each up to 80
В	TITLE2	characters in length (alphanumeric); see section 4.6
В	TLEAK	Total amount (g) of radioactive material leaked into
		a volume of contaminated air
В	V	Volume (cm ³) of contaminated air
В	VRATE	Air exchange rate (cm ³ /sec) for a volume (room) of
		contaminated air
B	X(K),Y(K),	Coordinates (cm) of a photon-emitting point source
	Z(K)	
В	X0,Y0,Z0	Coordinates (cm) of receptor
А	YI(J)	Abundance (g of mother/g of material) of the mother
		nuclide of the decay chain containing the J <u>th</u>
		daughter nuclide per gram of the radioactive material

Internal program variables

.

BCOR	Value of the Loevinger function (dimensionless);
	see section 3.1.3
CC(J)	Modified yield term = $YI(J)*RA(J)*CH(J)$; see section 3.1.3
F(ABS)	Finite volume correction factor (dimensionless); see
	section 4.2.1
FC	The value of c in the Loevinger function (dimen-
	sionless); see section 3.1.3
FNU	The value of $m{v}$, the apparent absorption coefficient
	(cm ² /mg) in the Loevinger function; see section 3.1.3
GAMCO	Dimensional factor = AVO*DEC(IEL)/3600

IFIRST Index variable used to determine if subroutine OUTPUT is being called for the first time in the evaluation of the product (dimensionless)

IGS Stage identification index (dimensionless)

- IRITE Flag variable (dimensionless) to prevent output of a nonexistent total group dose prior to entering the first group name in the output table
- ISP Current substage identification index (dimensionless)
- ISS Previous substage identification index (dimensionless)
- ISSP Current group identification index (dimensionless)
- ISSS Previous group identification index (dimensionless)
- ITAB Index variable (dimensionless) designating location of value being used in TABMU
- JLOC Location variable (dimensionless) used in reading in data
- PMEG Dimensional conversion factor (1.6E-8 rad-g/MeV X 1/2)
- STORGP Name (alphanumeric) of the stage considered prior to the current one
- STORSG Name (alphanumeric) of the substage considered prior to the current one
- STORSS Name (alphanumeric) of the group considered prior to the current one
- TERP1 Interpolation subroutines for obtaining photon linear TERP2 absorption coefficients; see section 4.4
- THETA Subroutine to evaluate THETA[EO(I)] in equation (3).
- XMU Photon linear absorption coefficient (cm⁻¹) in air given by TERP1 or TERP2

Output variables

DBETA	Beta	dose	rate	from	direct	contact	(rad/hr);	see
	equat	ion i	(3)					

DIR Dose equivalent rate associated with one year of product availability (mrem/year) for direct irradiation from an external photon source; see equation (1)

EM Dose equivalent rate associated with one year of product availability (mrem/year) for immersion in air contaminated with photon-emitting nuclides; see equation (2) EN 50-year dose commitment associated with one year of

product availability (mrem/year) from inhalation of radioactive material; see equation (5)

- ENG 50-year dose commitment associated with one year of product availability (mrem/year) from ingestion of radioactive material; see equation (4)
- GPDOS Contribution of an event dose equivalent rate to its
 group population dose rate (man-rem/year) = TOT*EVPROB*
 NM

GPMDOS Substage population dose equivalent rate (man-rem/year)
GTDOS Total population dose equivalent rate (man-rem/year)
STGDOS Stage population dose equivalent rate (man-rem/year)
TGPDOS Group population dose equivalent rate (man-rem/year)
TOT Total individual (event) dose equivalent rate associated
with one year of product availability (mrem/year)
= DIR + EM + EN + ENG

4.6 Description of Input

A description of the card input for CONDOS is presented in this section. This description and use of Fig. 12 will allow the user to take full advantage of the flexibility inherent in a computer code with a generalized format. Each rectangular box in Fig. 12 represents a data card.

The first four cards describe the problem and provide data for the beta dose calculation. The information contained in these cards need not be reentered.

The identification number (IEL) of the radioactive material is given first. CONDOS will treat only natural thorium (IEL=1) and natural uranium (IEL=2) at this time. Other materials can be considered if appropriate changes are made to the data base; that is, values must be supplied for each YI(J), RA(J), CH(J), CINH(J), CING(J), CIMM(J), ENRG(J), FBT(I,J), PGM(I,J), DEC(IEL), JHI(IEL), and JLO(IEL).

The second card contains the input data required for the calculation of DBETA; namely, SOB, ROB, STH, HTH, and TBETA. Cards 3 and 4 contain a brief description of the problem under study (TITLE1 and TITLE2). Both cards may contain up to 80 alphanumeric characters.

Point 1 is the first decision point; "is this the end of the input data?" If no more data are supplied, a blank card is inserted and data input is terminated. If more data are to be supplied, the stage index number (IGP=1,2,3,4,5) and the corresponding substage index number (ISP=1,2,...) are specified. The number of events (NOEV) that describe the group to be considered, the number of affected persons in the group (NM) and the name of the group (SSGRP) are also contained on this card.



Figure 12. Logical flow of input data card sequencing.

At point 3, the decision is reached as to whether or not a new stage is to be considered. If a new stage is to be considered, a card containing GROUP is entered. If not, no card is provided. Point 4 is a decision about substages which is similar to that for stages. If a new substage, enter SUBGRP; if not, omit card and go to point 5.

The name of the event (IVENT) to be considered next is supplied. The remaining cards provide the input information required for the dose calculations. Event specific data are provided by the next two cards: (1) NP=0,1,... K, ICON=1,2,3 (see section 4.2.2), ITN=0,1 (see point 10), EVPROB (a number between 0 and 1), R, TINH, X0, Y0, and Z0; (2) CONC (if ICON=1), TDIR, AMT, TIMM, SRATE (if ICON=2), TLEAK (if ICON=3), VRATE (if ICON=2), and V (if ICON=3).

If there are no external point sources (NP=0), point 8 in the flow sheet shows that the direct irradiation dose calculation is bypassed and point 14 is considered next. If NP >0, points 9 through 13 are considered for each source. Therefore, NP sets of these cards must be provided. S(K), X(K), Y(K), and Z(K) are provided.^a If ITN=0, AATN(K) must be provided, or, if ITN=1, ATN(I,K)--for each photon energy interval and point source--must be provided. Both AATN(K) and ATN(I,K) have values between 0 and 1.

When all point sources have been provided for, point 14 is considered. If the event just considered is not the last one in the group, decision point 5 is reconsidered and points 5 through 14 repeated for each event that describes the group. If the event just considered is

^aThree methods for coordinate specification are given in section 3.1.1. If any coordinate is to be set equal to zero, no data entry need be made -- a blank space will be interpreted as 0.

the last one for a group, decision point 1 is reconsidered. This last cycle of data card input is repeated until the data necessary to describe each event, group, substage, and stage have been supplied.

4.7 Description of Output

Subroutine OUTPUT generates a table of dose equivalent rates to representative individuals according to exposure mode and event. Doses from each mode are summed to provide the individual's total doses, and these are weighted to indicate the contribution from each event to the group population dose. Group, substage, stage, and total population dose equivalent rates to the total body are listed after all contributions to them have been listed individually.

The beta hand dose is listed separately and the problem description given by the user (TITLE1 and TITLE2) precedes the output table.

An output table from a sample problem is shown in Appendix C.

5. SUMMARY AND FUTURE DEVELOPMENTS

A methodology is described for estimation of potential radiation doses to man from the distribution, use, and disposal of consumer products that contain radioactive material. The methodology consists of two components: a model and a computer code. The model provides a format for identifying the parameters required to solve the dose equations, i.e., the conditions of exposure to the product. The computer code provides the solutions to the dose equations and lists the solutions in tabular form.

The format provided by the model divides the life span of a consumer product into five broad categories called stages. These stages

include distribution, transport, use, disposal, and emergencies. Since a stage can encompass a variety of population groups and activities of man, subdivision of the stages is required before attempting to estimate the rediation doses associated with them. This is accomplished by division of each stage into a series of substages, each of which may encompass several groups of persons such that all group members interact with the product in quite similar circumstances. The events during which group members may be exposed to the product are then identified and an individual from the group selected to take part in the event. The conditions of exposure for the selected individual are defined. These conditions are the parameters that are used in the solutions of the dose equations.

The present version of the CONDOS computer code provides solutions to standard dose equivalent rate equations and has a generalized format that allows a variable number of events, groups, substages, and stages to be considered. The doses potentially associated with one year of product availability to the total body of the selected individuals are computed according to the relevant pathways of exposure. Direct irradiation, immersion in contaminated air, ingestion, and inhalation pathways are considered. Summation of the doses to individuals over the proper population groups provides estimates of the population doses that will accrue to each group, substage, and stage population identified in the model. A beta skin dose is estimated also, for a specified duration of contact with the product by an individual.

The present version of the CONDOS computer code provides dose estimates to the total body from consumer products that contain either

natural uranium or natural thorium. The capability to estimate doses to other selected organs or tissues would increase the utility of the program, because the total body is not the critical organ for internal deposition of many radionuclides, including those of uranium and thorium. This capability will be incorporated in the next version of the computer code.

Many consumer products contain other radioactive materials, such as Pm-147 and H-3. The provision of a larger radionuclide data file would enable a larger variety of consumer products to be investigated. This is being done and should be available soon.

Incorporation of a standard event file would ease greatly the burden of data input that is placed on the user of the present version of CONDOS. Such an event file could not be prepared at the present time, but, as product evaluations are made, the information needed to construct such a file should accumulate. When data sufficient to make incorporation of such a file has accumulated, incorporation will be considered.

A point source representation is used in the estimation of the doses from direct irradiation. Many exposure situations can arise in which such a representation may not be as desirable as a line, surface, or volume source representation. Consideration will be given to providing the user with options for source representation.

CONDOS will be used to provide estimates of the radiation doses to man from consumer products that contain byproduct or source materials.¹ This work is in progress and will be reported periodically.

6. ACKNOWLEDGMENTS

The model and computer code reported in this work was developed from a basic plan provided by Dr. A. N. Tse of the Product Standards Branch, Directorate of Regulatory Standards, U.S. Atomic Energy Commission. In addition, Dr. Tse has provided valuable suggestions throughout the development of the methodology.

The assistance of the DRNL Information Center for Internal Exposure in providing unpublished radionuclide data and the valuable comments of Dr. P. S. Rohwer of the Environmental Sciences Division are also gratefully acknowledged.

REFERENCES

- Code of Federal Regulations, Title 10--Atomic Energy, Chapter I, Parts 20, 30-36, 49, and 70.
- K. Z. Morgan and J. E. Turner, eds., <u>Principles of Radiation Pro-</u>tection, John Wiley and Sons, Inc., New York, 1967.
- D. K. Trubey and S. V. Kaye, <u>The EXREM III Computer Code for Esti-</u> mating External Radiation Doses to Populations from Environmental <u>Releases</u>, ORNL-TM-4322, Oak Ridge National Laboratory, Oak Ridge, Tennessee (December 1973).
- 4. J. J. Fitzgerald, G. L. Brownell, and F. J. Mahoney, <u>Mathematical</u> Theory of Radiation Dosimetry, Gordon and Breach, New York, 1967
- J. A. B. Gibson and J. U. Ahmed, <u>Measurement of Short-Range Radia</u>-<u>tions</u>, Technical Reports Series No. 150, International Atomic Energy Agency, Vienna, 1973.
- 6. W. D. Turner, S. V. Kaye, and P. S. Rohwer, <u>EXREM and INREM Computer</u> <u>Codes for Estimating Radiation Doses to Populations from Construction</u> <u>of a Sea-Level Canal with Nuclear Explosives</u>, K-1752, Computing Technology Center and Oak Ridge National Laboratory, Jak Ridge, Tennessee (September 1968).

APPENDIX A

DATA BASE (JULY 1974)

The data base made available with the code in July 1974 includes the 17 nuclides that contribute about 99% of the dose from natural uranium and thorium which have been purified and aged for 20 years. Table A.1 lists the data variables related to individual radionuclides; the nuclides are also identified by name and index number J. The YI(J) are the natural abundance of the mother nuclides for each decay chain. The CH(J) are essentially the relative specific activities of the mothers to the principal nuclide in the uranium or thorium decay chains. The RA(J) are the relative disintegration rates of each nuclide in a decay chain to that of its mother. The RA(J) are time dependent terms in that they are determined at a specified time after purification of the radioactive material. These values were computed from the equations given in: I. Kaplan, <u>Nuclear Physics</u>, Addison-Wesley, Reading, Massachusetts, 1955.

The dose conversion factors CIMM(J), CINH(J), and CING(J) were supplied by the EXREM III³ and INREM⁶ computer codes, as noted in section 3.

All radionuclide data in the data base were assembled from:

1. C. M. Lederer, J. M. Hollander, and I. Perlman, <u>Table of Iso-</u> <u>topes</u>, sixth edition, John Wiley and Sons, New York, 1968.

Unpublished data of the Information Center for Internal Exposure,
 Oak Ridge National Laboratory.

BLANK PAGE

.

.

Table A.2 lists the values of FGM(I,J), the photon energy and yield per disintegration of each nuclide. Table A.3 lists the corresponding quantities for beta emission, FBT(I,J).

Table A.4 lists the data variables related to photon and beta particle energy. The energy interval index I is specified according to the range of energies included in each interval and the average photon [ENRGY(I)] and beta particle [EO(I)] energy in that range. The mean beta particle energy, EA(I), corresponding to EO(I) was obtained from M. F. James, B. G. Steel, and J. S. Story, <u>Average Electron Energy in</u> <u>Beta Decay</u>, AERE-M 640, Atomic Energy Research Establishment, Harwell, England, 1960. The exposure rate-energy flux ratios, CONVR(I), were obtained from H. Goldstein, <u>Fundamental Aspects of Reactor Shielding</u>, Addison-Wesley, Reading, Massachusetts, 1959.

Radionuclide CIMM(1) CINH(1) CINC(1) ENDE(1)								
Index J	Identifica- tion	YI (J)	RA(J)	CH(J)	<u>mrem/hr</u> µCi/cm ⁹	rem/µCi	rem/µCi	MeV
l ^{a,p}	Th-232	1.0	1.00	1.00	1.18E+3	I.84E+2	7.36E-2	0.015
2	Ra-228	1.0	0.874	1.00	3.21E+4	2.22E+1	1.66E+1	0.100
3	Ac-228	1.0	0.874	1.00	7.52E+5	1.57E-2	6.26E-6	1.000
4	Th-228	1.0	0.824	1.00	3.58E+3	4.16E+1	1.66E-2	0.015
5	Ra-224	1.0	0.824	1.00	1.02E+4	2.70E-1	2.02E-1	0.200
6	Rn-220	1.0	0.824	1.00	0.0	0.0	0.0	0.500
7	Po-216	1.0	0.824	1.00	0.0	3.46E-8	7.425-9	0.100
8	Pb-212	1.0	0.824	1.00	1.48E+5	1.10E-2	3.02E-3	0.200
9	Bi-212	1.0	0.824	1.00	1.11E+5	9.49E-4	3.65E-5	0.800
10	T1-208	1.0	0.297	1.00	3.42E+6	2.11E-6	1.98E-6	0.700
וו	Po-212	1.0	0.527	1.00	0.0	8,97E-14	1.92E-14	0.100
12 ^{a,b}	U-238	0.993	1.00	1.00	1.20E+3	1.17	4.50E-2	0.015
13	Th-234	0.993	1.00	1.00	7.57E+3	5.74E-3	2.30E-6	0.090
14	Pa-234m	0.993	1.00	1.00	2,81E+2	8.06E-5	3.23E-8	0.700
15 ^a	U-234	6.0E-5	1.00	1.87E+4	1.55E+3	1.33	5.13E-2	0.015
16 ^a	U-235	7.1E-3	1.00	6.48	1.56E+5	1.20	4.82E-2	0.200
17	Th-231	7.1E-3	1.00	6.48	2.97E+4	5.04E-5	2.02E-8	0.015

Table A.1 Data variables related to individual radionuclides.

^aMother nuclide of the decay chain of nuclides which follow it.

^bPrincipal nuclide in radioactive material.
I ,J=	l	2	3	4	5	6	7
	Th-232	Ra-228	Ac-228	Th-228	Ra-224	Rn-220	Po-216
1	0.0361			0.0442	0.00189		
2	0.0338		0.306	0.0414	0.00201		
3	0.00455		0.0199	0.00557			
4							
5							
6							
7	0.00191		0.00501				
8							
9				0.0129	0.00325		
10			0.0584	0.00058	0.00076		
11			0.0740	0.00014	0.00024		
12			0.0193	0.00298			
13			0.0563	0.00307	0.0401		
14			0.247				
15			0.0245				
16			0.0353			0.00067	
17							
18							
19			0.0784				
20			0.225				
21			0.197				
22			0.105				
23							
24							
25							

Table A.2	Photon yields per disintegration o	f each
	radionuclide (J) per energy interv	al (I).

•

FGM(I,J)

			F	FGM(I,J)			
I ,J=	8	9	10	11	12	13	14
	Pb-212	Bi-212	T1-208	Po-212	U-238	Th-234	Pa-234m
1	0.0660	0.0701	0.0124				
2	0.0702	0.00503	0.0132		0.0719	0.0413	0.000513
3					0.00468	0.00269	0.00034
4							
5		0.00897	0.0030				0.00001
6					0.00059		
7						0.00962	
8	0.0896	0.00136	0.0464				
9	0.162	0.00218	0.0106				
10	0.0748	0.00056	0.00305			0.0663	0.00115
11	0.00636				0.00028		0.00312
12							
13	0.462		0.00462				
14	0.0284	0.00439	0.0768				0.00047
15	0.00017	0.00015					
16		0.00395	0.234				
17			0.852				
18		0.0710					0.00354
19		0.00958					0.00384
20		0.00388	0.117				
21		0.00777	0.00598				
22		0.0208					
23							
24			0.998				
25							

-

Table A.2. Photon vields per disintegration of each radionuclide (J) per energy interval (I) (continued).

	FGM(I,J)				
I,J=	15	16	17	 	
	U-234	U-235	Th-231		
1					
2	0.0874	0.349	0.720		
3	0.00569	0.227	0.137		
4		0.00156	0.121		
5		0.00506	0.0005		
6	0.00161				
7			0.00454		
8			0.0024		
9			0.0864		
10		0.0849	0.0129		
11	0.0058	0.0488	0.0306		
12		0.151	0.00331		
13		0.584	0.0004		
14					
15					
16					
17					
18					
19		•			
20					
21					
22					
23					
24					
25					

Table A.2. Photon yields per disintegration of each radionuclide (J) per energy interval (I) (continued).

I,J=	1	2	3	4	5	6	7
	Th-232	Ra-228	Ac-228	Th-228	Ra-224	Rn-220	Po-216
1		1.00					
2							
3							
4							
5			0.105				
6			0.055				
7							
8							
9							
10			0.113				
11			0.065				
12			0.330				
13							
14							
15							
16							
17							••
18			0.200				
19							
20							
21			0.130				
22							
23							
24							
25							

Table A.3 Beta particle yields per disintegration of each radionuclide (J) per energy interval (I).

FBT(I,J)

67

I,J=	8	9	10	11	12	13	14
	Pb-212	Bi-212	T1-208	Po-212	U-238	Th-234	Pa-234m
1						0.190	
2	Ŭ.050					0.810	
3	0.810						
4							
5		0.008					
6	0.140	0.021					
7		0.013					
8							
9		0.001					
10			0.038				
11							
12							
13			0.250				0.0074
14							
15			0.210				0.0072
16		0.050					
17							
18			0.500				
19							
20							
21							
22							
23		0.547					0.985
24							
25							

Table A.3 Beta particle yields per disintegration of each radionuclide (J) per energy interval (I) (continued).

FBT(I,J)

			FBT(I,J)		
I,J=	15	16	17	 	····
	U-234	U-235	Th-231		
1			0.014		
2			0.277		
3			0.709		
4					
5					
6					
7					
8					
9					
10					
11					
12					
13					
14					
15					
16					
17					
18					
19					
20					
21					
22					
23					
24					
25					

.

Table A.3 Beta particle yields per disintegration of each radionuclide (J) per energy interval (I) (continued).

. .

Energy	Photon			Beta		
group index I	energy interval MeV	ENRGY(I) MeV	CONVR(1) <u>r/hr</u> MeV/cm ² -sec	energy interval MeV	EO(1) MeV	EA(I) MeV
1	0-0.012	0.01	2.901E-4	0-0.15	0.1	0.026
2	0.013-0.018	0.015	8.183E5	0.16-0.25	0.2	0.054
3	0.019-0.025	0.02	3.302E-5	0.26-0.35	0.3	0.084
4	0.026-0.035	0.03	0.955E-5	0.36-0.45	0.4	0.116
5	0.036-0.045	0.04	0.424E-5	0.46-0.55	0.5	0.149
6	0,046-0,055	0.05	0.259E-5	0.56-0.65	0.6	0.183
7	0,056-0,065	0.06	0.198E-5	0.66-0.75	0.7	0.217
8	0.066-0.075	0.07	0.169E-5	0.76-0.85	0.8	0.253
9	0.076-0.085	0.08	0.163E-5	0.86-0.95	0.9	0.289
10	0.086-0.095	0.09	0.154E-5	0.96-1.05	1.0	0.326
11	0.096-0.12	0.10	0.160E-5	1.06-1.15	1.1	0.363
12	0.13-0.18	0.15	0.173E-5	1.16-1.25	1.2	0.401
13	0.19-0.25	0.20	0.185E-5	1.26-1.35	1.3	0.442
14	0,26-0,35	0,30	0.199E-5	1.36-1.45	1.4	0.481
15	0.36-0.45	0.40	0.204E-5	1.46-1.55	1.5	0.522
16	0.46-0.55	0.50	0.204E-5	1.56-1.65	1.6	0,563
17	0.56-0.65	0.60	0.204E-5	1,66-1.75	1.7	0.605
18	0.66-0.75	0.70	0.202E-5	1.76-1.85	1.8	0.647
19	0.76-0.85	0.80	0.199E-5	1.86-1.95	1.9	0.688
20	0.86-0.95	0.90	0.196E-5	1.96-2.05	2,0	0.731
21	0.96-1.2	1.00	0.192E-5	2.06-2.15	2.1	0.774
22	1.3-1.7	1.50	0.177E-5	2.16-2.25	2,2	0.818
23	1.8-2.5	2.0	0.164E-5	2.26-2.35	2.3	0.862
24	2.6-3.5	3.0	0.145E-5	2.36-2.45	2.4	0.906
25	3.6-4.5	4.0	0.133E-5	2,46-2,55	2,5	0.950

.

Table A.4 Data variables related to photon and beta particle energy

APPENDIX B

DERIVATION OF THE MODEL EQUATION FOR CALCULATING

THE BETA DOSE RATE

The model equation used in CONDOS, Equation (3), is derived from the well-known Loevinger point-source dose-rate formula. 4,5 Details of the derivation of Equation (3) are provided in this Appendix. The Loevinger formula is of the form

$$L(x) = K (vx)^{-2} [cA_1(x) + A_2(x)]$$

where L(x) = beta dose rate (unspecified units, depending on K) at distance x in an absorbing medium from a point source of unit strength,

- K = normalization constant (to be dimensioned below),
- v = apparent absorption coefficient (cm²/g); = 18.6 (E₀ - 0.036)^{-1.37},
- $E_0 = maximum beta particle energy (MeV_:eta),$
- $x = distance (g/cm^2)$ travelled by the beta particle in the absorbing medium,
- c = parameter (dimensionless); $= \begin{cases} 2.0, \text{ if } 0.17 \le E_0 \le 0.5, \\ 1.5, \text{ if } 0.50 < E_0 \le 1.5, \\ 1.0, \text{ if } 1.50 < E_0 \le 3.0, \end{cases}$ $A_1(x) = \begin{cases} 0, \text{ if } x \ge c/v, \\ 1 (vx/c)exp(1 vx/c), \text{ if } x < c/v, \\ 1 vx exp(1 vx). \end{cases}$

The values of c and v given above are for tissue, e.g., the skin. In the formulation of the model equation, $E_0 = 0$ was set as the lower limit for c = 2; $E_0 = \infty$ was set as the upper limit for c = 1. Since only a few beta particles with energies below 0.1 MeV or above 3 MeV are emitted by nuclides in the natural uranium and thorium decay chains, this formulation will introduce a negligible error in practice.

The transformation function, in mass units (g/cm^2) , for conversion from a point kernel to an infinite plane kernel is

$$P(z) = \frac{2\pi}{\rho^2} \int_{z}^{\infty} x L(x) dx$$
, where ρ = density of the absorbing medium (g/cm³).

When transformed by this expression, the Loevinger formula becomes

$$P(z) = \frac{2\pi K}{\rho^2 v^2} \left\{ c \left[1 - \ln \frac{vz}{c} - exp(1 - \frac{vz}{c}) \right] + exp(1 - vz) \right\},$$

where $[] = 0$, if $z > c/v$.

The transformation, in mass units, from an infinite plane to a corresponding slab of volume sources for a receptor a distance s (g/cm^2) from a slab of thickness h (g/cm^2) is

$$D(s,h) = \rho^{-1} \int_{s}^{s} \frac{h}{P(z)} dz,$$

$$D(s,h) = \frac{2\pi K}{\rho^{3} v^{2}} \left\{ c \int_{s}^{s+h} [1 - \ln \frac{vz}{c} - \exp(1 - \frac{vz}{c})] dz + \int_{s}^{s+h} \exp(1 - vz) dz \right\}, (B.1)$$
where $[] = 0, \text{ if } s + h > c/v.$

CONDOS estimates beta doses only to the skin from contact with a consumer product. Therefore, equation (B.1) is solved for two cases: (1) when the beta particle range (c/v) is greater than or equal to the receptor distance (s) and less than or equal to the combined receptor distance and source thickness (s + h), and (2) when c/v is greater than s + h. The solutions are:

Case 1: $s + h \ge c/v \ge s$;

$$D(s,h) = \frac{2\pi K}{\rho^3 v^2} \left\{ c \int_{s}^{c/v} [1 - \ln \frac{vz}{c} - \exp(1 - \frac{vz}{c})] dz + \int_{s}^{s+h} \exp(1 - vz) dz \right\},$$

$$D(s,h) = \frac{2\pi K}{\rho^3 v^3} \left\{ c^2 [B_1] + [C_1] \right\}, \qquad (B.2)$$

where $[B_1] = 3 - \frac{vS}{c} (2 + \ln \frac{c}{vS}) - \exp(1 - \frac{vS}{c})$,

$$[C_1] = \exp(1 - vs) [1 - \exp(-vh)]$$

Case 2: s + h < c/v;

$$D(s,h) = \frac{2\pi K}{\rho^{3} v^{2}} \left\{ c \int_{s}^{s+h} [1 - \ln \frac{vz}{c} - \exp(1 - \frac{vz}{c})] dz + \int_{s}^{s+h} \exp(1 - vz) dz \right\},$$

$$D(s,h) = \frac{2\pi K}{\rho^{3} v^{3}} \left\{ c^{2} [B_{2}] + [C_{1}] \right\},$$
(B.3)

where
$$[B_2] = \frac{v(s+h)}{c} [2 + \ln \frac{c}{v(s+h)}] - \frac{vs}{c} [2 + \ln \frac{c}{vs}]$$

-exp $(1 - \frac{vs}{c})[1 - exp(-\frac{vh}{c})].$

The normalization constant K can be evaluated from the following considerations:

(1) The total energy deposited in a unit mass by beta particles, neglecting bremsstrahlung conversion losses, for a homogeneous distribution of sources in an infinite medium is simply the total beta energy emitted per unit mass. Therefore, the dose rate is

$$D_{\infty} = 1.60E-8 (S/p) \overline{E}$$
,

where D_{∞} = dose rate (rad/hr),

1.60E-8 = dimensional conversion factor (1.60E-6 erg/MeV) (0.01g-rad/erg),

$$S/\rho$$
 = source strength (beta-hr⁻¹/g),

 \overline{E} = average beta particle energy (MeV/beta).

(2) The dose rate from an infinite distribution of sources is twice the dose rate from a semi-infinite distribution, i.e., $D_{\infty} = 2D(o,\infty)$. Because $D(o,\infty)$ gives the dose rate at the surface of an infinite half space source, $D(o,\infty)$ can be evaluated by solving equation (B.1) with s = o and s + h = ∞ . This solution is

$$D(o,\infty) = \frac{2\pi K}{\rho^3 v^3} [3c^2 - e(c^2 - 1)].$$

From the above considerations,

1.60E-8 (S/p)
$$\overline{E} = \frac{4\pi}{\rho^{3}v^{3}} \Theta$$
, where $\Theta = 3c^{2} - e(c^{2}-1)$.
Therefore, $K = \frac{1.60E-8}{4\pi} \frac{(S/p)}{6} \overline{E} \frac{p^{3}v^{3}}{p^{3}v^{3}}$.

Substitution of the above value for K into equations(B.2) and (B.3) yields:

Case 1: D (s,h) = 0.80E-8 (S/
$$\rho$$
) $\overline{E} \ \Theta^{-1} \left\{ c^2 \ [B_1] + [c_1] \right\}$ (rad/hr); (B.4)

Case 2: D (s,h) = 0.80E-8 (S/
$$\rho$$
) $\overline{E} \odot^{-1} \left\{ c^2 [B_2] + [C_1] \right\}$ (rad/hr). (B.5)

Equations (B.4) and (B.5) are used in the CONDOS computer code. They are modeled for a source that emits beta particles of several energies. The source is represented in a manner consistent with section 3.1.1. The model equation is listed in section 3.1.3 as equation (3), and can be related to equations (B.4) and (B.5) by the following substitutions:

0.80E-8 = PMEG; $S/\rho = \left(\frac{SOB}{ROB}\right) *AVO*DEC(IEL)*FBT(I,J)*CC(J);$ $\overline{E} = EA(I);$ $\Theta^{-1} = THETA(EO(I));$ $\{ \} = BCOR(STH,HTH,EO(I),IAR), where IAR = 1 solves equation$ (B.4) and IAR = 2 solves equation (B.5); s = STH; and h = HTH. Equation (3) contains also the correction factors that allow the user to express STH and HTH in commonly used units, i.e., mg/cm^2 .

APPENDIX C

SAMPLE PROBLEM

The use of CONDOS is demonstrated by estimating the radiation doses to man from a hypothetical consumer product: earthen tableware designed for use in homes and commercial eating establishments. The tableware is assumed to be coated with a glaze containing 20 wt% natural uranium which has aged 20 years since purification. Since the density of the glaze is assumed to be 4.2 g/cm^3 , the concentration of uranium in the glaze is 0.84 g/cm^3 . The average piece of tableware is assumed to have a surface area of 750 cm² and to contain 100 g of uranium.

Shipments of the tableware are assumed to be restricted to long-haul and local-delivery truck. Warehouses and retail stores are the only distribution facilities assumed to contain the tableware. A single emergency situation is considered, i.e., a truck accident involving a cargo fire. Broken pieces of tableware are discarded in trash cans and disposed of in sanitary landfills.

Table C.1 lists and indexes the parts of the consumer product life span model (see section 2.2) selected for this example. Note that only a few of the possible parts were chosen. The names, as listed in the table, 0 [1] aloges; substages, groups, and events are the alphanumeric input statistics.

The input variables required to describe each event are listed in Table C.2. The first three lines of the table are the input data required by the first four input cards (see section 4.6). Even specific data are lines tabulated. Details of the formulation of this data will not be discussed, because the problem is hypothetical and no attempt was made Table C.1. Life span model for hypothetical tableware

1. Distribution 1.1 Warehouse 1.1.1 Handlers 1.1.1.1 handling 1.1.1.2 storage 1.1.2 Nearby residents 1.1.2.1 at home 1.1.3 Passersby 1.1.3.1 passing by 1.2 Retail stores 1.2.1 Clerks 1.2.1.1 display 1.2.1.2 storage 1.2.2 Shoppers 1.2.2.1 at display 1.2.2.2 in store 1.2.3 Nearby clerks 1.2.3.1 at work 1.2.4 Passersby 1.2.4.1 normal 2. Transport 2.1 Truck - long haul 2.1.1 Drivers - helpers 2.1.1.1 driving 2.1.1.2 stops - rest 2.1.1.3 accident - cargo fire 2.1.2 Persons at stops 2.1.2.1 normal activities 2.1.3 Persons along route 2.1.3.1 normal activities 2.1.4 Motorists 2.1.4.1 passing trucks 2.2 Truck - local delivery 2.2.1 Deliverymen 2.2.1.1 normal activities 2.2.2 Persons along route 2.2.2.1 normal activities 3. Use 3.1 Domestic use 3.1.1 Frequent users 3.1.1.1 eating 3.1.1.2 storage 3.1.2 Occasional users 3.1.2.1 eating 3.1.2.2 storage 3.1.3 Guests 3.1.3.1 eating

3.1.4 Neighbors

*

3.1.4.1 at home

3.2.1 Dishwashers 3.2.1.1 at work 3.2.2 Servers - waiters 3.2.2.1 at work 3.2.3 Patrons 3.2.3.1 dining out 3.2.4 Passersby 3.2.4.1 normal activities 4. Disposal 4.1 Landfill 4.1.1 Collectors 4.1.1.1 at work 4.1.2 Passersby 4.1.2.1 passing cans 4.1.3 Fill workers 4.1.3.1 at work 4.1.4 Area users 4.1.4.1 normal activities 5. Emergencies 5.1 Transport accident 5.1.1 Firemen 5.1.1.1 fighting fire 5.1.2 Policemen

3.2 Commercial use

5.1.2.1 crowd control

5.1.3 Medical team 5.1.3.1 first aid

IEL=2; TITLE1: TITLE2:	sob= Sam Con	0.84; PLE P TAINI	ROB ROBL	=4.2; STH= EM FOR CON 00 G OF 20	=40; 1005 D-YE/	hth=1 - hyi Ar oli	566; Pothi D NA	TBETA= ETICAL I TURAL U	1 Earthi Raniui	ENWARE 4 - HY	COA Pothi	TED ETIC	WITH AL PA	CERAI NRAME	MIC GLI TERS.	AZE										
Index	1GP	ISP	NOEV	NM	NP	ICON	ITN	EVPROB	R	TINH	X0 '	YO Z	0 c0	NC	TDIR	AMT	TIMM	SRATE	TLEAK	VRATE	۷	S(K)	X(K)	Y(K)	Z(K)	AATN(K)
1.1.1.1	1	1	2	1200	ĩ	<u> </u>	0	1.0	_						0.1						-	2.000	30.5			1.
1.1.1.2	-	-	-	-	j	j	Õ	1.0							40							2,000	305			i. –
1.1.2.1	1	1	1	1200	1	1	0	1.0							70							2,000	18,000			0.9
1.1.3.1	1	1	1	100000	1	1	0	1.0							0.02							2,000	9,000			0.9
1.2.1.1		2	2	500	ļ		U	1.0							120							500	150			0,9
1 2 2 1	ī	2	2	າມບຸບັນບາ	, 1	ł	ň	1.0							480							500	150			1.
1 2 2 2	-	-	-	-	i	í	ŏ	1.0							0.5							200	1.500			0.85
1.2.3.1	1	2	1	2000	i	i	ŏ	1.0							1920							400	18,000			0,5
1.2.4.1	1	2	1	999000000	i	1	0	1.0							0.02							400	18,000			0.5
2.1.1.1	2	1	3	510	1]	Õ	1.0							28							4,000	650			0.8
2.1.1.2	-	-	-	-	1		0	1.0	1000	0 00			1 00	10	8		0 00					4,000	9,000			0.6
2.1.1.3	-	7	ī	- 5100	- Ü	ł	0	3.5E-3	1000	0.08			1.2E-	10	0		0.08					4 000	0 000			0.6
2131	2	í	í	35700000	í	i	ŏ	1.0							0.02							4,000	15,250			0.3
2.1.4.1	2	i	i	35700	i	i	ŏ	1.0							0.02							4.000	3.050			0.6
2.2.1.1	2	ż	i	200	i	i	Ó	1.0							2							200	650			0.8
2.2.2.1	2	2	1	1000000	1	1	0	1.0							0.02							200	9,000			0.7
3.1.1.1	3	1	2	2000	4	1	0	1.0							700	5E-3						500	30.5			0.9
-	-	-	-	-	-	-	-	-	-	-				-	-	-	-	-	-	-	-	500	152.5	76 7		0.9
-	-	-	-	-	-	-	-	-	-	-				-	-	-	-	-	-	-	-	500	76.3	76 3		0.9
2112	-	-	•	-	-	ì	ō	10	-	-				-	2920	-	-	-	-	-	-	400	1.525	/0.5		0.8
31.2.1	3	ī	2	2000	4	i	ŏ	1.0							100	1E-3						500	30.5			0.9
-	-	÷.	_	-	-	<u>.</u>	-	-	-	-				-	-	-	-	-	-	-	-	500	152.5			0.9
-	-	-	-	-	-	-	-	-	-	-				-	-	-	-	-	-	-	-	500	76.3	76.3		0.9
-	-	-	-	-	-	-	-	~	-	-				-	-	-	-	-	-	-	-	500	76.3	76.3		0.9
3.1.2.2	-	-	-	-	1	1	0	1.0							2920							400	1,525			0.8
3.1.3.1	3	1	1	40000	4	1	0	1.0							1	1E-5						500	30.5			0.9
-	-	-	-	-	-	-	•		-	-	•			-	-	-	-	-	-	-	-	500	76 3	76 3		0.9
-	-	-	-	-	-	-	-	-	-	-				-	-	-	-	-	-	-	-	500	76.3	76.3		0.9
	-	,	,	-	1	1	0	1 0	-						2920							400	4.575			0.8
3.1.4.1	3	1	- ¦	0000		2	ň	10	95	2000					2000		2000	1E-9		1.2E6	3.6E6	375	90.0			1.
3.2.1.1	2	2	i	1000	i	2	ŏ	i.ŏ	95	250					1000		250	1E-9		1.2E6	3.6E6	500	152.5			1.
3,231	3	2	i	7500000	i	ī	ō	1.0							4	1E-4						500	61.0			0.9
3.2.2.4	3	2	i	93900000	i	i	Ō	1.0							0.02							1,000	18,000			0.5
4.1.1.1	4	Ĩ	1	30	1	1	0	1.0							1							100	90.0			0.7
4.1.2.1	4	1	1	100000	1	1	Q	1.0							0.02							500	0,000			0.7
4.1.3.1	4]	1	40	1	!	0	1.0							1 0							500	30,000			0.5
4.1.4.1	4	1	1	1000000	1	1	Ŭ	1.0	1000	1			1 25	.10	1						ļ	50.000	3.050			0.7
5.1.1.1	5	1	1	5	1	1	ň	1 0	1000	1			1 2F-	10	i						į	50,000	7,100			0.7
5121	5	1	ł	2 3	1	i	ŏ	1.0	1000	0.25			1.2E-	10	0.25						-	50,000	7,100			0.7
		•	•			•	-																			

Table C.2. List of input variables that describe the events in the life span of the hypothetical tableware^a

^aDashes (-) indicate variables and input cards that may not be filled-in. All blank variables are set equal to zero.

.

to verify the realism of the parameters used in the event descriptions. Reference to sections 4.5 and 4.6 will allow the interested reader to dimension the variables. Note that the formats of the data as listed in the table are not necessarily those required by the computer code.

Almost all events describe the conditions for direct photon irradiation from a single point source. The receptor was set at the origin for all cases, and the source to receptor distance defined by a single coordinate, X(K), for most events. Multiple point sources requiring more than one coordinate are described for events 3.1.1.1, 3.1.2.1, and 3.1.3.1. The option to specify an AATN(K), which may be regarded as a shielding factor, is selected for all direct irradiation events. If the ATN(I,K) option had been selected, 25 values of that variable would have been required for each direct irradiation event for which the option was used.

Immersion in a cloud of contaminated air is considered in events 2.1.1.3 (ICON = 1; CONC specified), 3.2.1.1, and 3.2 2.1 (ICON = 2; CONC = SRATE/VRATE). Ingestion of the radioactive material is considered only for users of the tableware--events 3.1.1.1, 3.1.2.1, 3.1.3.1, and 3.2.3.1. Inhalation is considered for truckers (2.1.1.3) and emergency workers (5.1.1.1, 5.1.2.1, and 5.1.3.1) in a truck-cargo fire situation by specifying the concentration of radioactive material in air. Inhalation by dishwashers (3.2.1.1) and servers - waiters (3.2.2.1) makes use of the calculated value of CONC.

Table C.3 is the computer output of the radiation doses to man for the conditions assumed for the distribution, transport, use, disposal, and emergency care of the hypothetical consumer product. The most affected population groups and individuals can be determined easily, as

can the contributions to the total dose by each exposure pathway considered in the hypothetical example. The beta skin dose from one hour of contact is listed at the end of the table.

	1	00SF P2R	I NDIAIGNAL J	A PALABD (MA B	1/Y R)	1404	ISOU NOILYTN	
		I NAS		INAL	TOTAL	,		-L 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
	LIRECT		1541411CN	INCESTION	T	PROBABILLTY	6800P 7052	TOTAL Dose
DISTRIBUTION							(AAN-BEM/TR)	(MAN- BEN/YR)
WA P EHO USE								
HANDLERS (1.2008 03 4EMBERS)								
44 NCLING Storace	7.99E-02 3.19E-01	0-0 0	0.0	0.0	7.992-02 3.192-01	1.00E 00 1.00E 00	9.58E-02 3.83E-01	
TOTAL ECSE H Nearey Pestents (1.2002 03 Members)	ANEL ERS							4. 795-01
AT HOME	1. 44 2-04	0.0	0.0	0*0	1. 44 E-04	1.00E 00	1.73E-04	
TOTAL DOSE N Fasserry (1.0002 og members)	EAREY FICTORATS							1. 73E-04
PASSING EY	1. E5E-07	0.0	0.0	0.0	1.65E-07	1-00E 00	1.65E-05	
TOTAL LOSE R	ASSERSEY							1.652-05
TOTAL I RFTAIL STORES	DOSE 4A3 EHOL	asi						4. 792-C1
CLERKS (5.0002 02 MEHBERS)								
nlsplat Stora ge	8.51E-01 6.55E-02	0°0	0.0	0.0	8-912-01 6-552-02	1.00E 00 1.00E 00	4_46E-01 3.27E-02	
TOTAL DOSE Cl Shoppers { 1.0002 06 Members}	SXUZT							4. 782-01
AT DISPLAY In Store	4. 33E-04 1. 40E-05	0-0 0	0.0	0°0	4.132-04 1.40 E-05	1.00E 00 1.00E 00	4.138-01 1.40F-02	
TOTAL BASE SF Nearby Clerks (2,0007 03 Members)	HOEPER S							4.27E-01

EOPULATICS RAFIATION COSF EPOP THE DISTRIBUTION, USE, AND DISPOSAL

Table C.3. Computer output for sample problem--hypothetical tableware

OF CONSUMENT FRODUCTS CONTAINING RADIOACTIVE MATERIALS

4 . 402-04	0.0	0.0	0.0	4.40E-04	1.002 00	8.80E-04	
NEABBY CIEBRS							8.802-04
4. 59E- 09	0_0	0.0	0.0	4.59E-09	1.002 00	4.58E-0%	
PASSERSET						r	4- 58E-03
DOSE RETAIL	STOBES						9. 11E-01
TOTAL DOSE	DISTRIBUT	TION					1.39E 00
7.882-02 8.202-05 0.0	0.C 0.0 7.C51-12	0.0 0.0 7.25E-G3	0.0 0.0 0.0	7.68 E-02 8.80 E-05 7.25 E-03	1.002 00 1.002 00 3.502~03	4.02E-02 4.49E-05 1.29E-05	
DRIVEFS - HELFER	5						4 . 022- 02
1. 10z-05	0_0	0.0	0.0	1.102-05	1.00E 00	5.612-05	
PEPSONS AT STOPS	;						5 . 61 2- 05
3.83E-08	0_0	0.0	0.0	3. 63E- 08	1.00E 00	1.37E-03	
PERSCHS JIONG BO	OTE						1.372-03
1.92E-06	0.0	0.0	0.0	1.928-06	1.DOE 00	6.84E-05	
otcfists							6.842-05
DOSE TRUCK	- ICNG HAUL						4 . 172- 02
2.E1E-04	0.0	0.0	C. 0	2.81F-04	1.002 00	5.632-05	
ELIVERYMEN							5.632-05
1.281-08	0.0	0.0	0.0	1.287-08	1.00E 00	1.282-05	
ERSCNS ALONG RO	DTE						1.285-05
	4. 40 2-04 NEARBY CIERKS 4. 59 2-09 PASSERSEY DOSE RETAIL TOTAL DOSE 7. 88 2-02 8. 60 2-05 0.0 PRIVEFS - HELEER 1. 10 2-05 PRIVEFS - HELEER 1. 10 2-05 PRIVEFS - HELEER 3. 63 1-08 PERSONS A1 STOPS 3. 63 1-08 PERSONS JIONG RO 1. 92 2-06 NOT CFISTS DO SE TRUCK 2. 61 2-04 SELIVERYMEN 1. 26 2-08 PERSONS ALONG RO	4. 402-04 0.0 NEARBY CIERKS 4. 592-09 0.0 PASSERSEY DOSE RETAIL STORES TOTAL DOSE DISTRIBU 7. 682-02 0.0 8. 602-05 0.0 0.0 7. 651-12 DRIVEFS - HELFERS 1. 102-05 0.0 PERSONS A1 STORE 3. 631-08 0.0 PERSONS A1 STORE 1. 922-06 0.0 NOTOFISTS DOSE TRUCK - ICNS HAUL 2. 612-04 0.0 PELIVERYMEN 1. 262-08 0.0 PERSONS ALONG RODIE	4.402-04 0.0 0.0 NEARBY CIERKS 4.592-09 0.0 0.0 PASSERSET DOSE RETAIL STORES TOTAL DOSE DISTRIBUTION 7.6802-05 0.0 0.0 0.0 7.052-12 7.252-03 DRIVERS - HELFERS 1.102-05 0.0 0.0 PRESONS A1 STORES 3.632-08 0.0 0.0 PRESCNS JIONG BOUTE 1.922-06 0.0 0.0 NOTCHISTS DOSE TRUCK - ICNG HAUL 2.612-04 0.0 0.0 PERSCNS ALONG ROUTE 1.262-09 0.0 0.0 PERSCNS ALONG ROUTE	4.402-04 0.0 0.0 0.0 0.0 NEARBY CIERNS 4.592-09 0.0 0.0 0.0 PASSERSET DOSE RETAIL STORPS TOTAL DOSE DISTRIBUTION 7.882-02 0.0 0.0 0.0 0.0 7.051-12 7.252-03 0.0 DRIVEFS - HELFERS 1. 102-05 0.0 0.0 0.0 0.0 DRIVEFS - HELFERS 3.631-08 0.0 0.0 0.0 DEPSCNS JIONG ROUTE 1.922-06 0.0 0.0 0.0 DEPSC TRUCK - LCNG HAUI 2.612-04 0.0 0.0 0.0 ELIVERYMEN 1.202-05 0.0 0.0 0.0 DEPSCNS JIONG ROUTE	4.402-04 0.0 0.0 0.0 4.402-04 MEABBY CLERRS 4.592-09 0.0 0.0 0.0 4.592-09 PASSERSEY DOSE RETAIL STORES TOTAL DOSE DISTRIBUTION 7.882-02 0.0 0.0 0.0 7.682-02 8.802-05 0.0 0.0 0.0 7.252-03 0.0 7.252-03 0.0 7.252-03 DRIVERS - HELFERS 1. 102-05 0.0 0.0 0.0 0.0 1.102-05 PERSONS A1 STCPS 3.635-06 0.0 0.0 0.0 0.0 1.922-06 NOTCRISTS DOSE TRUCK - LCNG HAUL 2.612-04 0.0 0.0 0.0 0.0 1.287-08 VERSCNS ALONG ROUTE	4.402-04 0.0 0.0 0.0 4.402-04 1.002 00 MEARBY CIEBRS 4.592-09 0.0 0.0 0.0 4.592-09 1.002 00 PASSERSET DOSE RETAIL STORES TOTAL DOEE DISTRIBUTION 7.882-02 0.0 0.0 0.0 7.882-02 1.008 00 0.00 7.652-03 1.008 00 0.0 7.252-03 3.502-03 DRIVEFS - HELFERS 1.102-05 0.0 0.0 0.0 1.102-05 1.002 00 PERSONS A1 STOPS 3.632-06 0.0 0.0 0.0 1.002 00 PERSONS FIONG ROUTE 1.922-06 0.0 0.0 0.0 1.922-06 1.002 00 DEFECHS FIONG ROUTE 2.612-04 0.0 0.6 0.0 2.617-04 1.002 00 PELIVERTAIN 1.225-09 0.0 0.0 0.0 1.297-08 1.002 00 PERSONS ALONG ROUTE	4.402-04 0.0 0.0 0.0 4.402-04 1.002 00 8.802-04 HEADED CLERKS 4.592-09 0.0 0.0 0.0 4.592-09 1.002 00 4.582-05 PASSERSET DOSE RETAIL STORES TOTAL DOSE DISTRIBUTION 7.682-02 0.0 0.0 0.0 7.682-02 1.008 00 4.492-02 8.602-05 0.0 0.0 0.0 7.252-03 1.002 00 4.492-05 0.0 7.252-03 1.002 00 4.492-05 0.0 7.252-03 1.002 00 5.612-05 PREVERS - HELEERS 1.102-05 0.0 0.0 0.0 1.102-05 1.002 00 5.612-05 PREVENS A1 STCPS 3.632-08 0.0 0.0 0.0 1.922-06 1.002 00 5.612-03 PREVENS JIONG ROUTE 1.522-06 0.0 0.0 0.0 1.922-06 1.002 00 6.842-05 NOTCRISTS DOSE TRUCK - LOWS HAUI 2.612-04 0.0 0.0 0.0 1.2817-04 1.002 00 5.632-05 TELITERYNH 1.225-09 0.0 0.0 0.0 1.2817-08 1.002 00 1.282-05

TOTAL	DOSE TRUCK-1	LCCAL DELIVER	T					6.91F-05
4 SN	- ASCO IVLUI	TEANSPOFT						4. 18E-02
DOMESTIC USF								
PREQUENT USERS (2.0002 03 MEMEERS)								
EATING Storagp	1.51E 02 1.49E-01	0-0 0-0	0-0	1.742-01 0.0	1.518 02 1.492-01	1-00E 00 1-00E 00	3.02E 02 2.98E-01	
TOTAL LCSE H Occasional Useps { 2.0002 03 Hembers}	FRECOENT CSERS							3 . 02E 02
EATING Storage	2.16F 01 1.49E-01	0°C 0°C	0-0	3.497-02 0.6	2. 16E 01 1.49E-01	1.00E 00 1.00E 00	4.32E 01 2.9AE-01	
TOTAL COSE C Guests (4.000e 04 Yembers)	OCCASICKAI USERS							4.35E 01
EATING	2. 16E-01	0-0	0.0	3. 49 E-04	2.16E-01	1.00E 00	B.64E 00	
TOTAL CCSE G Neigheops (A.000e 03 Meneers)	suests							8. 64E 00
AT HOME	1 . 66E-02	0-0	0-0	0.0	1.662-02	1.00E 00	1.33E-01	
TOTAL COSE N	IFIGHBCFS							1, 33E-01
TOTAL COMMERCIAL USE	DOSE DOMESTI	C USE						3.55E 02
DISHWASHERS (5.000E 02 MEMBERS)								
AT WORK	3.44E 01	1.13E-12	1.265-03	0.0	3.44E 01	1.00E 00	1.72E 01	
TOTAL COSE C Servers - Waiters (1.0002 03 Members)	LISENAS FEFS							1.72E 01
AT WORK	7.93F 00	1.42E-13	1.57E-04	0-0	7.93E 00	1-00E 00	7.93E 00	
TOTAL DOSE S PATRCNS (7.5002 06 MEMBERS)	SERVERS - WALTERS							7.93E 00
DINING OUT	1.602-01	0-0	0*0	3. 49 2-03	1.83E-01	1-00E 00	1.37E 03	
TOTAL COSE P Passersby (3.9902 07 Members)	ATFORS							1.37E 03
HOFHAL ACTIVITIES	1. 151-03	0-0	0-0	0.0	1.15E-08	1.002 00	1.152-03	

TOTAL DOSE PASSERSEY Total Dose Conhercial DSE								1. 15E-03
								1.40E 03
DISPOSAL	TOTAL DOSE	OSE						1.75E 03
LANDFILL								
COLLECTORS (3.000E 01 MEMBERS)								
AT BORK	3.21E-03	0_0	0.0	0.0	3.21E-03	1.00E 00	9.638-05	
TOTAL DOSE CO Passersby { 1.000e os members)	LIICTCIS							9.63E-05
PASSING CANS	2.31E-05	0-0	0.0	0.0	2.312-05	1.00E 00	2.31E-03	
TOTAL DOSE PA: FILL WORKERS (4.000E 01 MEMBERS)	SERSET							2.31E-03
AT WORK	1.151-06	0-0	0.0	0.0	1. 15 2- 06	1.00E 00	4.592-08	
TOTAL COSE FI Area Users (1.000e 06 yenders)	LI WCFKERS							4.592-09
NORMAL ACTIVITIES	8.25E-07	0.0	0.0	0.0	8.258-07	1.00E 00	8.258-04	
TOTAL DOSE ARI	EA USERS							8. 252-04
TOTAL COSE LANEFILL								3.23E-03
TOTAI DOSE DISFOSAL BMERGENCIES								3. 232-03
TRANSPORT ACCIDENT								
FIRENFN (5.000E 00 MERBERS)								
FIREPIGHTING	1.40E-03	8.81E-11	9.07E-02	0.0	9.212-02	1.00E 00	4.60E-04	
TOTAL DOST FIR Folicemen (2.0000 00 members)	REMEN							4 . 602-04
CROWD CONTROL	2.58E-04	8_81E-11	9-071-02	0.0	9.09 E-02	1.00g 00	1.822-04	
TCTAL EOSE PCI Medical Team (3.000e oo members)	ICENEN							1.822-04
FIBST AID	6.452-05	2.20 - 11	2.27E-02	0.0	2,27E-02	1.00E 00	6.82E-05	
TOTAL DOSE MED	JCAL TEAM							6.822-05

TOTAL COSE TRANSPORT ACCIDENT	7. 10E-04
TOTAL DOSE ENERGENCIES	7. 102-04
GRAND TCTAL COSE	1.75E 03

POR SKIN CONTACT OF 1.0 HGURS THE EETA DUSE IS 5.09E-02 RADS.

APPENDIX D

. .

LISTING OF THE CONDOS COMPUTER CODE

.

THE WARE OF THIS PROGRAM IS CONDOS. B LOCK DATA COMMONIALPHITITLE1(20), TITLE2(20), GROUP(5), SUBGRP(5), SSGRP(5), IVEN 1T(5), STORSS(5), STORSG(5), STORGP(5) COMMON/ILTG/IR, IW, ID, IFIRST, IGP, ISP, INCR, IH, NM, NP, IGS, ISS, ISSP, FICON, ITH, IR ITE, IDT, DT, LI, J LOC (2), IEL, NOEV A, NBINT, NGINT, J LO(2), JHI(2) COMMONIZEA LX ZDIR, EM, EN, ENG, TOT, EVEFCE, GEDOS, TOPDOS, TAEX(100), TAEMU 1 (50) , ENRG (17) , ENRG L, ETRGH, DENRG , GPMD OS , STODOS , GTD OS , XO , YO , ZO , TDIR 2, CONC, SRATE, VEATE, TEAK, V, X(10), Y(10), 2(10), S(10), ATN(10, 25), TIM %M_ELRGY(25) ,EA(25) ,EO(25) ,CONVR(25) ,YI(17) ,TAEMU1 (20) 3, CIMM (17), CINH (17), AMT, CING (17), F, XMU, ENRG L1, ENRGH1, DENRG1, TINH 4, AATI(10), TAEX1 (100), CC (17), CH(17), FA(17), 6F BET(25) FBT(25,17) FGM(25,17) 5,GAMCO, SOB, ROB, DEC(2), STH, HTH, TBETA, DEETA DATA RA/1.,2*,8736,8239,5*,8236,2965,5271,6*1./ DATA CH/14*1.,1.80924,2*6.476/,Y1/11*1.,3*,9928,C0006,2*.C071/ DATA CINH/184, 22.15, C1565, 41.6, 2697, C. 3.463E-8, C1095, \$9.49E-4,2.11E-0, 6.968E-14,1.17, C0574, 9.C62E-5,1.33,1.20,5.C41E-5/ DATA Q1 NG/. C738, 16.62, 6.2618-6, C1664, 2023, 0., 7.47E-9, C03072, \$3.65E ~5,1.978E -0,1.922E -14,.C450,2.296E -0,3.225E - 0,.C513, 6. C482.2. C17E-8/ DATA CIMM/1.18E3,3.21E4,7.52E5,3.58E3,1.02E4,C.,0.,1.48E5, & 1. 11E 5,3. 42E0, C.,1.2E3,7.57E3,281.,1.55E3,1.56E5,2.97E4/ CATA ENRGY/.01, C15, C2, C3, C4, C5, C6, C7, C8, C9, 1, 15, 2, 3, £.4,.5,.E,.7,.8,.9,1.,1.5,2.,3.,4./ DATA CONVR/29.C107E-5, 8.1833E-5,3.302E-5,.555E-5,.424E-5,.259E-5, 6.198E-5,.1686E-5,.1625E-5,.1544E-5,.1596E-5,.1728E-5,.1854E-5, ε. 1986E -5, 2037E -5, 2043E -5, 2037E -5, 2016E -5, 1991E -5, 1956E -5, 3, 1923E-5, 1758E-5, 1636E-5, 1453E-5, 1333E-5/ DATA E0/.1,.2,.3,.4,.5,.6,.7,.6,.9,1.,1.1,1.2,1.3,1.4,1.5,1.6, 81.7,1.8,1.9,2.,2.1,2.2,2.3,2.4,2.5/ CATA EA/. C26.. C54, C84, 116, 149, 183, 217, 253, 289, 326, 363, 8.401,.447,.481,.522,.563,.605,.647,.688,.731,.774,.818,.862, 8,906,.957 DATA FRT/25*C.,1.,28*C.,.105,.C55,3*C.,.113,.C65,.33,5*C.,.2,2*O. ε,, 13, 105*C.,. C5,. ε1, 2*C.,. 14, 23 *C.,. C08,. C21,. C13, C.,. C01, 6*C., \$. C5 ,6* C, ,. 547 ,11* C, ,. (38,2* C, ,. 25 ,C, ,. 21 ,2* C, ,. 5 ,5 7* C, ,. 19 ,. 81 , £35 + C.,. C074, 0.,. C072, 7 * C.,. 9854, 52 * C.,. C14, . 277, .709, 22 * C./ DATA FGM/. C3608, C338, CC455, 3*C., CG191, 44*C., 306, C199, 3*C., C £ r501,2+ C.,. C584,. C7396,. C193,. C563,. 2473,. C245,. C353,2+ C.,. C784, £, 225, 1969, 105, 3 *C, , C4419, C414, C0557, 5*C, , C1292, C0058, C001 \$4,.C0298,.C0307,12*C.,.C0189,.C0201,6*C.,.C0325,.C0076,.C0024,0., £, C4008,27*C, , C0067,34*C, , C6607, C7019,5*C, , C896, 16173, C748, 8. C0636, 0.,. 46153, C2847, C0017, 10*C., C7008, C0503, 2*C., C0897, &2*C.,.C0136,.C0218,.C0056,3*C.,.C0439,.C0015,.C0395,C.,.C7102,.C0 & 956,.C0383,.C0777,.C2081,3*C.,.C1243,.C1322,2*C.,.C03,2*C.,.C4637 8.0106, 00305, 2*0, 00462, 07682, 0, 23358, 85157, 2*0, 11651, 005 8 98, 2*0, 99789, 27*0, 07188, 00468, 2*0, 00059, 4*0, 00028, 15*0, 8, C4131, CO269, 3*C., CO962, 2*C., C6628, 16*C., C00513, C0034, 0., &. COOC1, 4*C.,. CO115, GO317, 2*C.,. COO47, 3*C.,. CO354, CO384, 7*C. &. C8736 , C0569 , 2* C. , C0161 , 4*C. , C0058 , 15*C. , 34911 , 2266 , C0156 , &. C0506 , 4*C. , C8487 , C4678 , 15126 , 58399 , 13*C. , 71977 , 13712 , \$, 12135 , 0005 , C. , 00454 , 0024 , 08636 , 0129 , 03058 , 00331 , 8.0004,12*0./ CATA DEC/.243E-16,.739E-16/,JLO/1,12/,JHI/11,17/, 6 NB1 NT/25/, NG1 NT/25/ D4T4 ENRC/.015,.1,1.,.015,.2,.5,.1,.2,.6,.7,.1,.015,.09,.7,.015, 8.7,.015/ DATA TAPMUZ 2.252E - 02, 2.571E~C2, 2.877E-02, 2.957E-C2, 12.975E-02, 2.963E-C2, 2.9305-02, 2.893E-C2, 2.846E-C2, 2 2.744E-C2, 2.547E-C2, 2. 2.601E-C2, 2.5998-02, 2,506E-02, 3564E+02, 2,5258-02, 2.482E-C2, 2.448E-C2, 2.410E-02, 2.3 4805-02, 2.285E-C2,

2.3495-02,

2.2046-02,

578-02,

2.312E-C2,

2.159E -02,

2.181E = C2,

2.259E -C2,

2.132E -02,

2.22

2.112

C

2.C93E-C2, 6E - C2 2. C75E - C2, 2.0535-02, 2.C37E-C2, 2.C20E 7-02, 2.CO4E-62, 1,985E-02, 1.955E+C2, 1.970E-C2, 1.9412 -1.916E-02, 1.686E-C2, 1,8988-02, 802, 1,928E-C2, 1.875E-C 1,863E-C2, 1.852E-C2, 1.842E-C2, 1.831E-C2, 1.816E-C2 92, %/ 4.560E 00, DATA TAEMU1/ 4,9045-01, 1.412E-C1, 6.290E-C2, 2.920E-C2, A 3, 600E - C2, 2.364E-C2, 2.570E-C2. 2.320E-02. 82.312E-02, 2.328E-02, 2.358E-C2, 2.400E-02. 2.444E-C2, 2 C . 495E - 02 , 2.636E-C2, 2.527E-02, 2.561E - C2, 2.5985 - 02, 2. 2674E-C2/ DATA ENRO L/O. 10/, ENRGH/5.C/, DENRG/C. 10/, INCR/C/, 10/1/1/ 5, IW/6/, LPT/6/, IR/5/, ID/1/, ENRGL1/. C1/, ENRGH1/.2/, DENRG1/.01/ DATA JLOC/1,2/ F ND COMMONIALPHIT IT LET (20), TIT LEZ (20), GROUP(5), SUEGRP(5), SSGRP(5), IVEN 1T(5),STORSS(5),STORSG(5),STORGP(5) COMMON/INTG/IR, IW, ID, IFIRST, IGP, ISP, INCR, IH, NM, NP, IGS, ISS, ISSP, IICON, ITN, IR ITE, IDI, UT, LI, JLOC (2), IEL, NOEV A, NEINT, NGINT, JLO(2), JHI(2) COMMON/REALX/DIR, EM, EN, ENG, TOT, EVPPOB, GPDOS, TGPDOS, TABX(100), TABMU 1 (50), ENRG (17), ENRG L, ENRGH, DENRG, GPMDOS, STGDOS, GTDOS, XO, YO, ZO, TDIR 2, CONC, SRATE, VRATE, TLEAK, V, X (10), Y (10), Z(10), S(10), ATN(10, 25), TIM *M, ENRGY (25), EA (25), EO (25), CONVR (25), YI (17), TABMU1 (20) 3, CIMM (17), CINH (17), AMT, CING (17), F, X MU, ENRG LI, ENRGH1, DENRG1, TINH 4, AATN(10), TAEX1 (100), CC (17), CH(17), RA(17), LFBET(25), FBT(25,17), FGM(25,17) 5, GAMCO, SOB, FOB, DEC(2), STH, HTH, TBETA, DBETA IFIRST=0 GTDOS=0 CALL INPUT CALL IFILE(ID, 'INPU') FEAD(ID,103) IEL PEAD (ID, 104) SOB, ROB, STH, HTH, TBETA 104 FORMAT(5F12.4) CALL PREFAS READ(10,102) (TIT LE1(11),11=1,20) 102 F OR MAT (2044) READ(10,102) (TIT LE2(12),12=1,20) CALL OUTPUT IRITE=0 103 F OR MAT(15) 68 DO 71 LS=1.5 71 STORSS(LS)=SSGRP(LS) READ(10,100) IGP,15P,NOEV,NM,(SSGRP(LG),LC=1,5) 100 FORMAT(315,115,5X,5A4) IF(IGP) 51,52,51 51 IF (IGS-IGP) 54,53,54 54 DO 72 LS=1,5 72 STORGP(LS)=GROUP(LS) PEAD(ID, 105) (GROUP(LG), LG=1,5) 105 FORMAT(5A4) 53 IF(ISS=ISP) 57,56,57 56 IF(IGS-IGP) 57,59,57 57 DO 73 LS=1,5 73 STORSG(LS)=SUBGRP(LS) FEAD(ID,105) (SUBGRP(LG),LG=1,5) 59 DO 61 1=1,NOEV READ(10,105) (IVENT(LV), LV=1,5) FEAD (ID, 110) NP, ICON, ITN, EVPROB, E, TINH, XO, YO, ZC 110 FOR MAT(315, E12.2, F10.4, F7.2, 3F12.6) PEAD (ID, 115) CONC, TDIR, AMT, TIMM, SRATE, TLEAK, VRATE, V FORMAT(E10,2,E10.3,4E10.2,2E10.3) 115 IF(NP) 58,62,58 58 DO 62 J=1,NP READ(ID, 120) S(J), X(J), Y(J), Z(J)

С

```
120 FORMAT(E12.4,3F10.4)
      1= (ITN) 66,63,66
  63 DO 64 L=1, NGI NT
  64 ATN(J, L)=1.
      FEAD(10,150) AATN(J)
  150 FORMAT (F10.4)
      GO TO 6?
  66 READ(10,125) (ATN(J,L),L=),NG1NT)
  125 FORMAT(8F10.4)
      AATN(J)=1.
  62 CONTINUE
      IF ITE=IRITE+1
      CALL CALC
      CALL OUTPUT
  61 CONTINUE
       [SSP=1SSP+1
      GO TO 68
  52 WRITE(14,130) (STORSS(L), L=1,5), TGPDOS
130 FORMAT(1HC,10X,'TOTAL DOSE --- ',5A4,73X,1PE12.3)
       WRITE(IW, 135) (SUBGRP(L), L=1,5), GPMD OS
  135 FORMAT(1HC, 20X, 'TOTAL DOSE --- ',5A4,63X,1PE12.3)
       WRITE(IW, 140) (GROUP(L), L=1,5), STGDOS
  140 FORMAT(1HD, 30X, 'TOTAL DOSE --- ',5A4,53X,1PE12.3/)
       WRITE(IW, 145) GTDOS
  145 FORMAT (1H), 40X, 'GRAND TOTAL DOSE', 62X, 1 PE12.3)
       WEITE(14,00)TBETA ,08ETA
   90 FORMAT(1H1, 'FOR SKIN CONTACT OF ', F9.1, ' HOURS
& THE BETA DOSE IS', E14.7, ' RADS.'/)
       FERLIND 1
       CALL RELEAS(ID)
       STOP
       E ND
       SUBROUTINE CALC
      COMMON/ALPH/TITLE1(20),TITLE2(20),GROUP(5),SUBGRP(5),SSGRP(5),IVEN
      1T(5), STORSS(5), STORSG(5), STORGP(5)
      COMMON/INTG/IR, IV, ID, IFIRST, IGP, ISP, INCR, IH, NM, NP, IGS, ISS, ISSP,
     TICON, ITN, IRITE, IDT, LPT, LI, J LOC (2), IEL, NOEV
     A, NBINT, NGINT, JLO(2), JHI(2)
      COMMON ZREALX ZDIR, EM, EN, ENG, TOT, EVPROB, GPDOS, TGPDOS, TABX(100), TARMU
      1 (50) , ENRO (17) , ENRO L, ENROH, DENRO, GPMDOS, STODOS, GTDOS, XO, YO, ZO, TDIR
     2, CONC, SRATE, VRATE, TLEAK, V, X (10), Y (10), Z(10), S(10), ATN(10, 25), TIM
     *M , E NRGY (25) , EA (25) , EO (25) , C ONVR (25) , YI (17) , TABMUT (20)
     3, CIMM (17), CINH (17), A MT, CING (17), F, XMU, ENRGLI, ENRGHI, DENRGI, TINH
     4, AATN(10), TABX1 (100), CC (17), CH(17), FA(17),
      6FBET(25), FBT(25,17), FGM(25,17)
     5, GAMCO, SOB, ROB, DEC (2), STH, HTH, TBETA, DBETA
       JLOT = JLO(IEL)
       JHI1=JHI(IEL)
       D18=0
       IF(NP) 3,4,3
    3 SU MD = O
       DO 1 LAF1,NP
       SG M=0
       DO 2 LG=1, NGI NT
    2 SGM=SGM+FBET(LG)*ATN(LA,LG)
    1 SU MD # SU MD + S( LA) * GAMC O* SG M* AAT N( LA) / ((XO + X( LA)) * * 2
      #+(Y0-Y(LA))**2+(ZC-Z(LA))**2)
       DIR=SUMD*TDIR*79.6
    4 CONTINUE
C COMPUTE THE DOSE DUE TO IMMERSION (EM) .
      GO TO (8,9,11), ICON
     9 CONC=SRATE/VRATE
       GO TO 8
   11 CONC = T LEAK /V
```

```
8 EM=0
```

C

```
E N=O
      F NG = 0
       DO 7 L1=J LO1,JH11
      E M=E M+CC(L1)*C|MM(L1)*F(A BS)
      EN=EN+CINH(L1)+CC(L1)
    7 ENG=ENG+CING(L1)+CC(L1)
      E M=E M*C ONC *T | MM*GA MCO/(3,7E4 *8766.)
      EN=EN*CONC*TINH*.833E 09*GAMCO/3.7E4
      ENG = ENG + A HT + GA MC 0/37 .
       TOT=D IR +EM+EN+ENG
       GPDOS = TOT * EVPROB * NM / 1000 .
       GTDOS = GTDOS + GPDOS
       RETURN
      E ND
       FUNCTION F (AES)
      COMMON/ALPH/TITLE1 (20), TITLE2 (20), GROUP(5), SUBGRP(5), SSGRP(5), IVEN
     1T(5),STORSS(5),STORSG(5),STORGP(5)
      COMMON/INTG/IR, IW, ID, IFIRST, IGP, ISP, INCR, IH, NM, NP, IGS, ISS, ISSP,
     11 CON, ITN, IR ITE, ID1, LPT, L1, J LOC (2), IEL, NOEV
     A, NBINT, NGINT, JLO(2), JHI(2)
      COMMON/REALX/DIR, EM, EN, ENG, TOT, EVPROB, GPDOS, TGPDOS, TAEX(100), TAEMU
     1 (50) , ENRG (17) , ENRG L, ENRGH, DENRG , GPMDOS, STGDOS, GTDOS, X0, Y0, Z0, TDIR
     2, CONC, SRATE, VRATE, TLEAK, V, X(10), Y(10), Z(10), S(10), ATN(10, 25), TIM
%M, ENRGY(25), EA(25), EO(25), CONVR(25), YI(17), TAEMU1(20)
     3, CIMM(17), CINH(17), AMT, CING(17), R, XMU, ENRGLI, ENRGHI, DENRGI, TINH
     4 ,AATN(10) ,TABX1 (100) ,CC(17) ,CH(17) ,RA(17) ,
     6FBET(25), FBT(25,17), FGM(25,17)
     5,GAMCO,SOB,ROB,DEC(2),STH,HTH,TBETA,DEETA
       IF (ENRG ( L1 ), LT. C. C1. OR. ENRG ( L1 ). GT. 5.) GO TO 4
       IF(INCR) 2,1,2
     1 CALL TERP1
       GO TO 3
    2 CALL, TERP2
    3 F =1. -EXP(-X MU+R)
      GO TO 5
    4 WRITE(IW,200)
  200 FORMAT( THO ,5X, 'ENRG VALUE OUT OF LIMITS OF CROSS-SECTION TALLE')
    5 RETURN
       E ND
       SUBROUTINE TERPI
      COMMON/ALPH/TITLE1 (20), TITLE2 (20), GROUP (5), SUBGRP (5), SSGRP (5), IVEN
      1T(5),STORSS(5),STORSG(5),STORGP(5)
      COMMON/INTG/IR, IW, ID, IFIRST, IGP, ISP, INCR, IH, NM, NP, IGS, ISS, ISSP,
      11 CON, ITN, IR ITE, ID1, LPT, L1, J LOC (2), IEL, NOEV
      A, NBINT, NGINT, JLO(2), JHI(2)
      COMMON/REALX/DIR, EM, EN, ENG, TOT, EVPROB, GPDOS, TGPDOS, TAPX(100), TAPMU
      1 (50) , ENRC (17) , ENG L , ENG H , DENRG , G PMDOS , STGDOS , GTDOS , XO , YO , ZC , TD IR
     2, CONC, SRATE, VRATE, TLEAK, V, X(10), Y(10), Z(10), S(10), ATN(10, 25), TIM
#M, ENRGY (25), EA (25), EO (25), CONVR (25), YI(17), TAEMUI (20)
      3, CIMM(17), CINH(17), A MT, CING(17), R, X MU, E NRGL1, E NRGH1, DE NRG1, TINH
      4, AATN(10), TABX1 (100), CC (17), CH(17), FA(17),
      6FBET(25), FBT(25, 17), FGM(25, 17)
      5,GAMCO, SOB, POB, DEC (2), STH, HTH, TBETA, DPETA
C ENRG=GAMMA RAY ENERGY IN MEV.
  ENRGLELOWEST ENERGY VALUE IN TAPLE, TABMU.
   DENRG=ENERGY INCREMENT BETWEEN POINTS IN TABMU.
       IF (ENRG( L1).GT.ENRG L) GO TO 13
       TAB LOC = (E NRG ( L1 )-E NRG L1 )/DE NRG1
       GO TO 14
    13 TA5 LOC = (E NRG ( L1 )-E NRG L) /DF NRG
   14 ITAB=IFIX(TABLOC)+1
       FR=TABLOC+1.-ITAB
       IF (ENRG (L1) GT .ENRGL) GO TO 16
       XMU= (TABMU1 (ITAB+1)-TABMU1 (ITAB))*FR+TABMU1 (ITAE)
       GO TO 17
```

C

C

```
16 XMU= (TAEMU (ITAE+1)-TAEMU (ITAE))*FR+TAEMU (ITAE)
```

```
SUFROUTINE TERP2
    COMMONZALPHZTITLE1(20),TITLE2(20),GROUP(5),SUBGRP(5),SSGRP(5),IVEN
   IT(5), STORSS(5), STORSG(5), STORGP(5)
    COMMON/INTG/IR, IV, ID, IFIRST, ICP, ISP, INCR, IH, NM, NP, IGS, ISS, ISSP,
   IICON, ITN, IRITE, IDI, LPT, LI, JLOC (2), IEL, NOEV
   A,NBINT,NGINT,JLO(2),JHI(2)
    COMMONITE A LX / DIR, EM, EN, ENG, TOT, EVPROB, GPDOS, TGPDOS, TAEX (100), TABMU
   1 (50) , ENRG (17) , ENRG L, ENRGH , DENRG , G PMDOS, STG DOS, G TDOS, XO, YO, ZO, TD IR
   2, CONC, SRATE, VRATE, TLEAK, V, X(10), Y(10), Z(10), S(10), ATN(10, 25), TIM
   %M ,E NRGY (25) ,EA (25) ,EO (25) ,C ONVR (25) ,YI(17) , TABMU1 (20)
   3, CI MM (17), CI NH (17), A MT, CI NG (17), F, X MU, E NRG L1, E NRG H1, DE NRG1, TI NH
   4, AATN(10), TAEX1 (100), CC(17), CH(17), FA(17),
   6FBET(25), FET(25, 17), FGM(25, 17)
   5 , CAMCO , SOB , FOE , DEC (2), STH ,HTH , TBETA , DEETA
     10=0
     IF(ENRG(L1).GT.ENRGL) GO TO 1
  4 L0 = L0 + 1
     IF(TABX1 (LO)-ENRG(L1)) 4,5,6
  5 XMU=TAEMU1 (LO)
    GO TO 7
  6 FR= (TABX1 (LO) - E NRG (L1)) / (TABX1 (LO) - TABX1 (LO-1))
     XMU=TABMU1(LO)-FR*(TABMU1(LO)-TABMU1(LO-1))
    GO TO 7
  1 LO= LO+1
     IF (TABX(LO)-ENRG(L1)) 1.2.3
  2 XMU=TAE MU (LO)
    60 10 7
  3 FR= (TABX(LO)-ENRG(L1)) /(TAFX(LO)-TABX(LO-1))
     XMU = TAR MU(LO) - FR*(TAR MU(LO) - TAR MU(LO-1))
  7 RETURN
     E NO
     SUPROUTINE OUTPUT
    COMMONIALPHITITLE1(20), TITLE2(20), GROUP(5), SUBGRP(5), SSGRP(5), IVEN
   1T(5), STORSS(5), STORSG(5), STORGP(5)
    COMMON/INTG/IR, IW, LD, IFIRST, IGP, ISP, INCR, IH, NM, NP, IGS, ISS, ISSP,
   FICON, ITN, IRITE, IDI, LPT, LI, JLOC (2), IEL, NOEV
   A, NEINT, NGINT, J LO(2), JHI(2)
    COMMONIZEA LX / DIR , E M, E N, E NG , TOT , EV PROE , G PDOS , TG PDOS , TAEX(100) , TAEMU
   1 (50) , E KRG (17) , E KRG L, E KRG H, DE KRG , G PMD OS , STG DOS , STDOS , XO , YO , ZO, TD IR
   2 , CONC, SRATE, VRATE, TLEAK, V, X(10), Y(10), Z(10), S(10), ATN(10, 25), TIM
   5Å,ENRGY(25),EA(25),EO(25),CONVR(25),YI(17),ŤABMU1(20)
3,CIMM(17),CINH(17),AMT,CING(17),Γ,XMU,ENRGL1,ENRGH1,DENRG1,TINH
   4 ,AATN(10) , TABX1 (100) ,CC (17) ,CH(17) ,FA(17) ,
   6FBET(25), FPT(25, 17), FGM(25, 17)
   5 ,GAMCO, SOB , HOB ,DEC(2), STH, HTH, TBETA ,DEETA
     IF(IFIRST) 2,1,2
  1 WRITE (1W,900)
900 FORMAT(1H1,34X, 'POPULATION RADIATION DOSE FROM THE DISTRIBUTION, US
1E, AND DISPOSEL'/ THO, 39X, 'OF CONSUMER PRODUCTS CONTAINING RADIOACTI
    OVE PATERIALS(77)
     WRITE (1W ,905 )(T (T LE1(11),11=1,20)
905 FORMAT(1H ,25X,2044)
     WRITE(1,1,906) (TIT LE2(12),12=),20)
```

```
906 F CR MAT ( 1H , (25x, 20A 4) //)
     WRITE(IW,910)
```

17 RETURN E ND

```
910 FORMAT(46X, 'DOSE PER INDIVIDUAL INVOLVED (MREM/YR) 1,18X. COPULATION
   1 005E1)
```

```
WRITE(IW,911)
             WRITE(14,915)
```

```
915 FORMAT( 38X, 'EXTER NA L', 16X, 'INTER NA L', 11X, 'TOTA L')
1----!/)
     WRITE(IW ,920)
920 FORMAT( 33X, 'DIRECT', 5X, 'IMMERSION', 2X, 'INHA LATION', 3X, 'INGESTICH',
   115X, PROBABILITY', 7X, GROUP', 9X, TOTAL')
     WRITE(1W,919)
919 FORMAT(110X, 'DOSE', 10X, 'DOSE'/1H ,105X, '(MA N-FEM/YR)',2X, '(MA N-PEM
   1/YE) 1
    IFIRST=1
     165 = -1
     155 = - 1
    1555=-1
    GO TO 50
  2 CONTINUE
    1F(1GS-1GP) 4,3,4
  4 IF(IGP=1) 11,5,11
11 WRITE(1W,60C) (STORSS(L),L=1,5),TGPDOS
600 FORMAT(1+0,1CX,'TOTAL DOSE ---- ',5A4,73X,1FE12.3)
     WRITE(1W, 605) (STORSG(L), L=1,5), GPMD 05
605 FORMAT(1H0,20X, 'TOTAL DOSE --- ',5A4,63X,1PE12.3)
WRITE(1W,610) (STORGP(L),L=1,5),STGDOS
61C FORMAT(1H0,30X,'TOTAL DOSE ----',5A4,53
                                         ',5A4,53X,1PE12.3)
  5 WRITE(1W,925) (GROUP(L),L=1,5)
925 FORMAT(1H ,(5A4)/)
     STG DOS =0
  3 1 F(1SS-1SF) 7,6,7
  6 | F(1GS-1CP) 7,55,7
  7 IF(ISP-1) 13,12,13
 13 IF(IGS-IGP) 12,19,12
 19 WRITE(1W,600) (STORSS(L), L=1,5), TGPDOS
     WRITE(IW,605) (STORSG(L),L=1,5),GPMDOS
 12 WRITE(IW,930) (SUBGRP(L), L=1,5)
930 FORMAT(1H ,2X,(5A4)/)
    G PMD OS =0
 55 IF (ISSS-ISSP) 9,8,9
  9 (F(IRITE-1) 15,14,15
 15 IF(ISS-ISP) 14,25,14
 25 (F(IGS-IGP) 14,26,14
 26 WRITE(IW,600) (STORSS(L), L=1,5), TGPDOS
 14 WRITE(IW,935) (SSGRP(L),L=1,5)
935 FORMAT(1H ,4X,5A4)
     XM= NM
     WRITE(IV,936) XM
936 F OR MAT ( 1H ,5 X, '( ',1 PE 10.3, 1 X, 'ME MBERS) '/)
     ISSS=ISSP
     IGS=IGP
     ISS=ISP
     TG PDOS =0
   8 TG PDOS = 1G PDOS +G PDOS
     GPMDOS = CPMDOS+GPDOS
     STO DOS =STG DOS +G PDOS
     WR (TE(14, 940) (IVENT(1), L=1,5), DIR, EM, EN, ENG, TOT, EVPROB, GPDOS
 940 FORMAT(1H ,6X,5A4,1PE12.2,2X,1P4E12.2,1PE14.2,1PE13.2)
  50 RETURN
     E ND
     SUBROUTINE INPUT
     COMMON/ALPH/TITLE1(20), TITLE2(20), GROUP(5), SUBGRP(5), SSGRP(5), IVEN
    1T(5), STOR SS(5), STOR SG(5), STOR GP(5)
COMMON/INTG/IR, IW, ID, IFIRST, IGP, ISP, I NOR, IH, NM, NP, IGS, ISS, ISSP,
    11 CON, ITN, IRITE, ID1, LPT, L1, J LOC (2), IEL, NCEV
    A, NBINT, NGINT, JLO(2), JHI(2)
     COMMON/REA IX / DIR , EM, EN, ENG, TOT , EVPROB, GPDOS, TGPDOS, TAEX(100), TAEMU
    1 (50), ENRG (17), ENRG LE NRGH, DENRG, GPMDOS, STGDOS, GTDOS, XO, YO, ZO, TDIR
```

0,00%0,5RATE,VEATE,TLEAK,V,X(10),Y(10),Z(10),S(10),ATN(10,25),TIM *#,E12GY(25) ,E4 (25) ,EC (25) ,CONVR (25) ,Y1(17) ,TAEMU1 (20) 3, CI MY (17), CI TH (1/), AMT, CI NG (17), F, XAU, E NRG LI, E NRGH 1, DE NRG1, TI NH 4 ,AATN(1C) ,TAPX1 (100) ,CC (17) ,CH(17) ,FA(17) , 6FPET(25), FPT(25, 17), FGM(25, 17) 5 GANCO, SOB, FOF , DEC (2), STH, HTH, TBETA, DEETA CALL TEILE (IR, THEUT) 105 = -1135 = -1FEAD (IR ,403) IEL 403 FORMAT(15) WRITE (101,403) IEL RITE (LPT,404) IEL 404 FOPMAT(1H ,15) FEAD (IR, 406) SON, FOR, STH, HTH, TBETA 406 FORMAT(5F12.4) WRITE(ID1,406) SCB, FCB, STH, HTH, TBETA WRITE (UPT ,407) SOE , FOE ,STH ,HTH , TEETA 407 FORMAT(1H ,5F12.4) FEAD (IR ,400) (T IT LE 1 (L1), L1=1,20) VEITE(101,400) (TIT LE1(L1),L1=1,20) MOITE(LPT,401) (TITLE1(L1),L1=1,20) 400 FORMAT(2044) 401 FORMAT(1H ,20A4) FEAD (19,400) (T IT LES (LS), LS=1,20) WRITE(101,400) (TITLE2(L2), L2=1,20) WRITE (LPT ,401) (TIT LE2 (L2), L2=1,20) 68 READ(IR, 100) (GE, ISP, NOEV, NM, (SSGRP(LG), LG=1,5) 100 FORMAT(315,115,5X,5A4) WRITE(1D1,100) 10P,1SP,NOEV,NM,(SSORP(LG),LG=1,5) WEITE(LPT, 101) IGP, ISP, NOEV, NH, (SSGPP(LG), LG=1,5) 101 FORMAT(1H ,315,115,5X,5A4) IF(16P) 51,52,51 51 IF(ICS-IGF) 54,53,54 54 READ(IR,105) (GROUP(LG), LG=1,5) 105 FOR MAT (5A 4) VEITE(101,105) (GROUP(LG), LG=1,5) ARITE(LPT, 100)(0ROUP(LC), LC=1,5) 106 FORMAT(1H ,544) 53 (F(185-13P) 57,56,57 56 IF (165-16P) 57,59,57 57 READ (IR, 105) (SUBGRP(LG), LG =1,5) WHITE(101,105) (SUPERP(LG), LG=1,5) WRITE(LPT, 106)(SUEGRP(LG), LG = 1,5) 105 =1 6 P 135 =1 SP 59 00 61 1=1,NCEV FEAD(12,105) (IVENT(LV), LV=1,5) WRITE(101,105) (IVENT(LV), LV=1,5) WEITE(LPT, 10b) (IVENT(LV), LV=1,5) FEAD(ID, 110) NP, ICON, ITN, EVPROB, F, TINH, X0, Y0, Z0 110 FORMAT(315,E12, 2,E10, 4,E7,2,3E12.6) WRITE(ID1,110) NP, ICON, ITN, EVPROP, F, TINH, X0, Y0, 20 WRITE (LPT, 111) NF, ICON, ITN, EVPROB, F, TINH, X0, Y0, 20 111 FORMAT(1H ,315,6E12.3) FEAD (IR, 115) CONC, TDIR, AMT, TIMM, SRATE, TLEAK, VRATE, V 115 FORMAT(E 10.2, E10.3, 4E 10.2, 2E10.3) KRITE(101,115) CONC, TDIR, AMT, TIMM, SRATE, TLEAK, VRATE, V KRITE(LPT, 116) CONC, TBIR, AMT, TIMM, SRATE, TLEAK, VRATE, V 116 FORMAT(1H ,E10.2,E10.3,4E10.2,2E10.3) IF(NP) 58,62,58 58 DO 62 J=1,NP FEAD(IR, 120) S(J), X(J), Y(J), Z(J) 120 FOR MAT(E12.4,3FT0.4) IRITE(101,120) S(J),X(J),Y(J),Z(J)

٢.

```
WRITELEPT, 121) S(J), X(J), Y(J), Z(J)
121 FORMAT(1H ,E12.4,3E12.3)
     IF(ITN) 66,63,66
 66 READ(1R,125) (ATN(J,L), L=1, NGINT)
125 FORMAT(8 F10.4)
     wRITE(ID1,125) (ATN(J,L),L=1,NGINT)
     WRITE(LPT, 126) (ATN(J,L), L=1, NGI NT)
126 F OR MAT(1H ,8F10.4)
     CO TO 62
 63 READ (IR ,130) AATN(J)
130 FOR MAT(F10.4)
     WRITE(ID1,130) AATN(J)
     WRITE (LPT, 131) AATN(J)
131 FORMAT(1H ,F10.4)
 62 CONTINUE
 61 CONTINUE
     GO TO 68
 52 REWIND 1
  52 CALL RELEAS(IR)
     PETURN
     E ND
     SUBROUTINE PREFAS
    COMMON/ALPH/TITLE1(20), TITLE2(20), GROUP(5), SUBGREUS), SSGRP(5), IVEN
    1T(5), STORSS(5), STORSG(5), STORGP(5)
    COMMON/INTG/IR, IW, ID, IFIRST, IGP, ISP, INCR, IH, NM, NF, IGS, ISS, ISSP, IICON, ITN, IRITE, ID1, LPT, L1, JLOC(2), IEL, NOEV
   A, NBINT, NGINT, JLO(2), JHI(2)
     COMMON /REA LX / DIR, EM, EN, EN, ENG, TOT, EVPPOB, GPDOS, TGPDOS, TAEX(10C), TABMU
    1 (50) , E NRG (17) , E NRG L, E NRG H, DE NRG , G PMDOS, STG DOS, G TDOS, XO, YO, ZC, TD IR
    2, CONC, SRATE, VRATE, TLEAK, V, X(10), Y(10), Z(10), S(10), ATN(10, 25), TIM
    #M, E NRGY (25) , EA (25) , EO (25) , CONVR (25) , Y I (17) , TAF MU1 (20)
    3, CIMM(17), CINH(17), AMT, CING(17), F, XMU, ENRGLI, ENRGHI, DENRGI, TINH
    4, AATN(10), TAEX1 (100), CC (17), CH(17), FA(17),
    6FBET(25),FET(25,17),FGM(25,17)
    5,GAMCO, SOB, ROB, DEC (2), STH, HTH, TBETA, DEETA
     DIMENSION BCON (25)
     J LO1 = J LO (I E L)
     JH I 3 = J H I (I E L)
     CO 60 L=1,NBINT
 10 ECON(L)=1.
     PMEG=.5+1.6E-8
     AV0=.6023E24
     COEF=PMEG+AVO+DEC(IEL)+SOB/ROB
     COGAM=AVO+DEC(IEL)
     GAMCO=COGAM/3600.
     CO 1 L=J LO1 JHII
   1 CC(L)=Y1(L)+RA(L)+CH(L)
     FSM=0.
     DO 2 L=1,NBINT
     FBET(L)=0.
     00 3 LA = J LO1 , J H 11
   3 FBET(L) = FBET(L) + FET(L, LA) + CC(LA)
     IF (ST H+ HT H-FC (EO (L) ) /F NU (EO (L) )) 4,4,5
   4 IAR=2
     GO TO 6
   5 |AR=1
   6 FBET(L)=FEET(L)+EA(L)+BCOR(IAR,EO(L),STH,HTH)
    S*T HETA (BO (L)) *BCON (L)
   2 FSM=FSM+FBET( L)
      DBETA = FSM+COEF+TBETA
     DO 7 L=1, NGI NT
     FBET(L)=0.
      DO 8 LA=J LO1 ,JH11
```

C

8 FBET(L) = FEET(L) + FGM(L,LA) + CC(LA) FBET(L) = FBET(L) +E NRGY (L) +C ONVF (L) 7 CONTINUE FETURN E ND FUNCTION BCOR(IR,E0,S,H) 81=F NU (EO) 82=FC (EO) B3=B1+S P4 =6 3/8 2 85=81+H E1=EXP(1,-B3) IF(85-60.) 10,11,11 11 854 =60. GO TO 12 10 E54 #85 12 CONTINUE E2=1,-EXP(-854) E3=82#82 E4 =1. - A LOG (84) E5=2XP(1,-84) IF(IR-1)1,1,2 1 BCOR=E 1*E2+E3*(3,-B4*E4-E5) GO TO 50 2 B 6=B1 + (S+ H) /B7 E(=2. -A LOG (36) B7=85/82 IF (37-60.) 20,21,21 21 B 7A=60. GO TO 22 20 E7A=B7 22 CONTINUE E 7=1, -EXP(-B 7A) BCOR=E1+E2+E3+(B6+E6-B4 +E4 -E5+E7) 50 RETURN E ND FUNCTION THETA (EC) THETA =1./(EXP(1.)+FC(EC)++2+(3.+EXP(1.))) FETURN ENU FUNCTION FC (EG) IF(, 5-E0)2,1,1 1 FC=2. GO TO 50 2 1F(1, 5-EC) 4,3,3 3 FC=1.5 GC TO 50 4 FC=1. 50 RETURN E ND FUNCTION FNU (EO) EX=-1.37 FNU=18.6E-3*(E0-.C36)**EX DETURN E ND